Contents lists available at ScienceDirect

Computer Networks



journal homepage: www.elsevier.com/locate/comnet

Survey paper

A survey on ambient backscatter communications: Principles, systems, applications, and challenges $\stackrel{\star}{\approx}$

Weiqi Wu, Xingfu Wang*, Ammar Hawbani, Longzhi Yuan, Wei Gong

University of Science and Technology of China, Hefei, 230026, China

ARTICLE INFO

Keywords: Ambient backscatter Backscatter communications Ambient backscatter communications systems Low-power communications Internet of Things Energy harvesting

ABSTRACT

Owing to the nature of ultra-low-power and extremely low deployment as well as maintenance costs, ambient backscatter technology is promising to become a prominent choice of future low-power communications systems, especially Internet of Things (IoT). Considerable studies have been reported in the emerging area of ambient backscatter communications. However, there is a lack of this kind of survey that is specific to a certain area of ambient backscatter communications and discusses some of the latest systems and the developments in the emerging area. Therefore, in this paper, we provide a comprehensive survey of the existing literature related to ambient backscatter communications. We first present the basic principles of ambient backscatter communications and primer knowledge of ambient signals. After that, we provide a new taxonomy for ambient backscatter communications systems proposed in the literature for each category. Then, we describe potential applications driven by ambient backscatter communications. Finally, some open issues and challenges for future research are identified and discussed.

1. Introduction

In recent years, the significant advances made in modulated backscatter technology and energy harvesting (EH) technology spurred the development of backscatter communications (BackComs). In BackComs systems, the backscatter transmitter (i.e., the tag), which is a batteryfree device, can harvest energy from incident signals sent from a nearby excitation source (i.e., a carrier emitter) and transmit its data just by modulating and reflecting the incoming excitation signals [1–3]. In this way, BackComs eliminate the need for wires and batteries, thereby reducing the power consumption and promising to achieve ubiquitous communication. In general, BackComs can be broadly categorized into the following three types:

(i) Monostatic BackComs (Mon-BackComs): This type of BackComs uses a dedicated carrier emitter. The carrier emitter and the backscatter receiver are equipped inside the same device called the reader. Owing to the co-located configuration, the reflected signal experiences a round-trip path loss [4]. In addition, this configuration suffers from the doubly near-far problem [5]. Specifically, far tags from the reader can experience a less amount of the harvested energy and a lower signal strength than near tags [6]. As a result, the system performance is limited in terms of communication range and throughput. Moreover, Mon-BackComs inevitably incur installation and maintenance costs due to the use of a dedicated carrier emitter.

- (ii) Bistatic BackComs (Bis-BackComs): The carrier emitter and the backscatter receiver are separated in this type of BackComs. Thus, the round-path loss can be avoided in Bis-BackComs. Furthermore, the doubly near-far effect can be alleviated by deploying carrier emitters near backscatter tags. However, Bis-BackComs also need to deploy and maintain special excitation sources, thereby increasing the cost.
- (iii) Ambient BackComs (Amb-BackComs): This type of BackComs is similar to Bis-BackComs. Different from Bis-BackComs, the signals (e.g., Wi-Fi and TV) transmitted in existing systems can act as the excitation signals in Amb-BackComs. Therefore, Amb-BackComs eliminate the need to allocate a dedicated frequency spectrum and install a special carrier emitter, thereby greatly improving frequency spectrum resource utilization and reducing deployment and maintenance costs.

* Corresponding author.

https://doi.org/10.1016/j.comnet.2022.109235

Received 25 April 2022; Received in revised form 30 June 2022; Accepted 24 July 2022 Available online 30 July 2022 1389-1286/© 2022 Elsevier B.V. All rights reserved.



 $[\]stackrel{ imes}{\sim}$ This work was supported by NSFC, China Grant No. 61932017 and 61971390.

E-mail addresses: wwqacmy@mail.ustc.edu.cn (W. Wu), wangxingfu@ustc.edu.cn (X. Wang), anmande@ustc.edu.cn (A. Hawbani), longzhi@mail.ustc.edu.cn (L. Yuan), weigong@ustc.edu.cn (W. Gong).

 Table 1

 Abbreviations used throughout the paper

Abbreviation	Description	Abbreviation	Description		
EH	Energy Harvesting	IoT	Internet-of-Things		
BackComs	Backscatter Communications	CSS	Chirp Spread Spectrum		
AC	Alternating Current	DC	Direct Current		
BLE	Bluetooth Low Energy	GFSK	Gaussian Frequency Shift Keying		
MAC	Medium Access Control	OQPSK	Offset Quadrature Phase-Shift Keying		
Mon-BackComs	Monostatic Backscatter Communications	DSSS	Direct-Sequences Spread Spectrum		
ISM	Industrial Scientific Medical	FPGA	Field-Programmable Gate Array		
Amb-BackComs	Ambient Backscatter Communications	FFT	Fast Fourier Transform		
DBPSK	Differential Binary Phase Shift Keying	DQPSK	Differential Quadrature Phase Shift Keying		
OFDM	Orthogonal Frequency Division Multiplexing	MIMO	Multiple-Input Multiple-Output		
LoRa	Long Range	Bis-BackComs	Bistatic Backscatter Communications		
SC-FDMA	Single Carrier-Frequency Division Multiple Access	PSS	Primary Synchronization Signal		
SSS	Secondary Synchronization Signal	BER	Bit Error Ratio		
PER	Packet Error Ratio	ADC	Analog-to-Digital Converter		
P-FSK	Pseudo-Frequency Shift Keying	OOK	On-Off Keying		
S-BPSK	frequency-Shifted Binary Phase Shift Keying	BPSK	Binary Phase Shift Keying		
DAC	Digital-to-Analog Converter	VCO	Voltage-Controlled Oscillator		
CSI	Channel State Information	ACK	Acknowledgment		
RS	Reed–Solomon	TV	Television		
FM	Frequency Modulation	Wi-Fi	Wireless Fidelity		
LTE	Long-Term Evolution	UE	User Equipment		

The remarkable advantages in Amb-BackComs technology make it possible to play a key role in future Internet-of-Things (IoT) applications [7-10]. Significant studies have been reported in the emerging area of Amb-BackComs. Therefore, a survey paper is needed to review the existing literature related to Amb-BackComs and to provide a clear sight for future research efforts in this direction. There have been several outstanding surveys and tutorials [11–16] in the field of BackComs. Huynh et al. [11] provide a detailed survey of Amb-BackComs. The paper provides a detailed discussion on the fundamentals of modulated backscatter communications. In addition, the paper also presents some backscatter systems and discusses some challenges that the Amb-BackComs systems face. Xu et al. [12] provide a survey of BackComs. The paper provides a discussion on the backscatter basics and different modulation methods, and presents some Bis-BackComs systems as well as some potential research directions. Memon et al. [13] present a detailed survey on the aspects of leveraging BackComs to solve the limited battery life problem in emerging networks. Muratkar et al. [15] provide a detailed discussion on various methodologies proposed in the literature to make battery-less wireless communication a reality. Rezaei et al. [16] present a detailed survey of the integral aspects of wireless-powered networks and BackComs.

While there exist survey articles on BackComs, these articles do not provide a discussion on the primer knowledge of ambient excitation signals. It is very important for readers to understand the operation and development of Amb-BackComs systems. In addition, these articles focus on the whole field of BackComs. There is a need for a survey paper that is specific to a certain area of Amb-BackComs and discusses some of the latest systems and the developments in the emerging area. Furthermore, most of the existing surveys discuss Amb-BackComs systems from energy efficiency, throughput, communication range angles. However, most of the existing Amb-BackComs systems are designed specifically for particular ambient signals. Therefore, a new taxonomy for Amb-BackComs systems from ambient signals angles is expected to provide readers with a clearer insight into the developments of Amb-BackComs systems.

Motivated by the aforementioned observations, we provide a new survey paper. The contributions of this survey are summarized as follows:

(1) This paper provides a discussion on the primer knowledge of different ambient signals, including Bluetooth, Wi-Fi, ZigBee, TV, FM, LTE, and LoRa. The discussed primer knowledge, not found in other papers, is of great importance for readers to have a basic understanding of the operation and development of Amb-BackComs systems.

(2) This paper provides an exhaustive discussion on different Amb-BackComs systems that have been proposed in the literature between 2013 and 2021. We provide a summary of different Amb-BackComs systems from the performance perspective, including minimum power, maximum throughput, and maximum range.

(3) This paper provides a new taxonomy for Amb-BackComs systems, based on the type of ambient signals (e.g., Bluetooth or Wi-Fi). This is because most of the existing Amb-BackComs systems are designed specifically for particular ambient signals. This new taxonomy can provide readers with deep visibility into the development of Amb-BackComs systems.

(4) This paper also discusses some practical applications empowered by Amb-BackComs, and different challenges that Amb-BackComs currently face.

The rest of this paper is organized as follows. Section 2 presents the principles of Amb-BackComs including architecture, basic techniques, and primer knowledge of ambient excitation signals. Section 3 provides a taxonomy for Amb-BackComs systems. We also provide a detailed discussion on different Amb-BackComs systems proposed in the literature for each category. Section 4 describes applications driven by Amb-BackComs. Open issues and future research directions are discussed in Section 5. Finally, Section 6 concludes this paper. The abbreviations in this article are summarized in Table 1.

2. Principles of Amb-BackComs

In this section, we first present a typical Amb-BackComs system architecture. Then, basic techniques used in Amb-BackComs systems are discussed. Finally, we provide a discussion on the primer knowledge of ambient signals.

2.1. Architecture

An Amb-BackComs system typically consists of three main components, i) ambient sources, ii) backscatter tags, and iii) backscatter receivers, as shown in Fig. 1 (a) [11,17,18]. The backscatter tag conveys its data to the backscatter receiver by backscattering ambient signals. In this paper, ambient signals include Bluetooth, Wi-Fi, ZigBee, TV, FM, LTE, and LoRa. A commodity device (e.g., a Wi-Fi device) or another backscatter tag can act as the backscatter receiver. The backscatter tag is a battery-free device, which consists of an antenna module, an energy harvester and power management module, a detector module, a processing subsystem module, and a load modulator module, as shown in Fig. 1(b).

· Antenna: to transmit and receive radio waves.



Fig. 1. (a) A typical Amb-BackComs system architecture. (b) A general backscatter tag architecture.



Fig. 2. A typical energy harvesting architecture [19], in which AC and DC are short for alternating current and direct current, respectively.

- *Energy harvester and power management*: to harvest energy from ambient signals and ambient light, and to power the device.
- Detector: to detect the excitation signal.
- *Processing subsystem*: to control load modulator to generate backscattered signals.
- *Load modulator*: to modify the characteristics (amplitude, frequency, and phase) of the incoming ambient signals in order to transmit data.

2.2. Basic techniques used in Amb-BackComs systems

2.2.1. Energy harvesting (EH)

The fundamental purpose of energy harvesting is to harvest energy from ambient signals and ambient light, and provide energy for the device. The energy harvesting module is very important for passive tags. A basic EH module consists of two main components: i) an energy transducer component and ii) a power management component, as shown in Fig. 2 [19]. The energy from ambient signals or ambient light is first converted into usable electricity through the transducer. However, the output of the transducer is highly unstable, which cannot be used directly to power the device, since conventional electronic components are designed for stable direct current (DC) supply [19]. Thus, the transducer is followed by a power management module that is used to further regulate the electricity generated by the transducer in order to provide a stable power supply for the device. In general, the power management component includes a rectifier that converts an alternating current (AC) to a direct current, a DC-to-DC converter that converts an unstable DC to a stable DC, and a storage element that stores the energy.

2.2.2. Detector

The detector is a key component on the backscatter tag for detecting the presence of the excitation signal. Once the signal is detected, the backscatter tag will perform a series of operations to piggyback its data on the signal. Traditional detector generally requires power-hungry components, which are not feasible for low-power tags. The existing detector used on the tag can be broadly classified into two categories: i) leveraging the strength of the incoming signal and a threshold voltage to detect the presence of a specific signal, and ii) leveraging the features of a specific signal to detect the presence of the signal. Zhao et al. [20] present a design of a Wi-Fi signal detector, which falls into the first category. As shown in Fig. 3(a), the detector consists of two key modules: i) a multistage demodulating logarithmic amplifier and ii) a voltage comparator. First, on the backscatter tag side, the incoming signal is amplified and summed up by a multistage amplifier circuit. Then, a lowpass filter is used to remove unwanted noise. After that, the voltage comparator circuit compares the amplifier's output (denoted by V_{RF}) with a threshold voltage (denoted by V_{ref}). The comparator will output a high voltage if $V_{RF} > V_{ref}$, and a low voltage otherwise. This high voltage implies that a Wi-Fi signal is detected. The experiment results show that this detector can clearly identify the beginning, duration, and ending of Wi-Fi signals.

Peng et al. [21] present a design of other category of the detector, as shown in Fig. 3(b). This detector leverages the features of LoRa preamble to identify LoRa signals. Peng et al. [21] notice that the LoRa preamble consists of ten identical up chirps. Thus, we can identify the existence of LoRa signals by correlating the incoming signal with the pre-stored up chirp. The experiment results demonstrate that this detector can achieve more than 99% accuracy for LoRa signal detection over a distance of 50 m.



Fig. 3. (a) Design of a Wi-Fi signal detector circuit [20], and (b) Packet detection workflow on the PLoRa tag [21].

2.2.3. Modulated backscatter

Modulated backscatter technique is based on the fact that a radio wave when encountered a boundary between two media with diverse impedance or densities will be reflected back, and the amount of reflection usually depends on the variation in impedance or density values [1]. Thus, by tuning the load impedance of the antenna, a backscatter tag can piggyback its data on the backscattered signal. Specifically, let $S_{in}(t) = A_{in}e^{j(2\pi f_{int}t+\theta_{in})}$, $S_{out}(t) = A_{out}e^{j(2\pi f_{out}t+\theta_{out})}$, and Γ_{tag} be the excitation signal, the backscattered signal, and the reflection coefficient of the antenna, respectively. Thus, the backscattered signal $S_{out}(t)$ is computed by Eq. (1) [12,22]:

$$S_{out} = S_{in}(t) \times \Gamma_{tag}$$

= $A_{in} e^{j(2\pi f_{in}t + \theta_{in})} \times \Gamma_{tag}$ (1)

In Eq. (1), Γ_{tag} can be expressed as follows [12,23]:

$$\Gamma_{tag} = \frac{Z_l - Z_a}{Z_l + Z_a} = \left| \Gamma_{tag} \right| e^{j\theta_{tag}}$$
(2)

In Eq. (2), $Z_a = |Z_a|e^{j\theta_a}$ and $Z_l = |Z_l|e^{j\theta_l}$ are an antenna impedance and a load impedance, respectively. $|\Gamma_{tag}|$ and θ_{tag} are computed as follows:

$$\begin{aligned} \left| \Gamma_{tag} \right| &= \frac{\left| Z_{a} \right|^{2} + \left| Z_{l} \right|^{2} - 2 \left| Z_{a} \right| \left| Z_{l} \right| \cos(\theta_{a} - \theta_{l})}{\left| Z_{a} \right|^{2} + \left| Z_{l} \right|^{2} + 2 \left| Z_{a} \right| \left| Z_{l} \right| \cos(\theta_{a} - \theta_{l})} \end{aligned} \tag{3} \\ \theta_{tag} &= \arctan(\frac{2 \left| Z_{a} \right| \left| Z_{l} \right| \sin(\theta_{a} - \theta_{l})}{\left| Z_{a} \right|^{2} - \left| Z_{l} \right|^{2}}) \end{aligned}$$

Based on Eq. (2), Eq. (1) can be written as:

$$S_{out} = A_{in} e^{j(2\pi f_{in}t + \theta_{in})} \times \Gamma_{tag}$$

= $|\Gamma_{tag}| A_{in} e^{j(2\pi f_{in}t + \theta_{in} + \theta_{tag})}$ (5)

From Eq. (3)–(5), we can conclude that the backscattered signal S_{out} can only be modified by changing the value of the load impedance Z_l , since the impedance of a given antenna, i.e., Z_a , is constant. Thus, by switching among a set of loads, we can change the amplitude and phase of the backscattered signal accordingly. To do this, the tag uses a *switch* that connects the antenna to a set of loads and toggles between different loads. The *switch* is controlled by the processing subsystem (e.g., microcontroller or *field-programmable gate array* (FPGA)). It is worth noting that the frequency of the backscattered signal can be modified by changing the frequency of toggling *switch* [22]. As a result, a tag can adjust one or more of the components, i.e., amplitude, frequency, and phase, of the backscattered signal to transmit its data to a backscatter receiver.

2.3. Primer knowledge of ambient signals

The selection of ambient signals is significant in designing Amb-BackComs systems and the transmission efficiency of Amb-BackComs systems usually depends on the type of ambient signals (e.g., Wi-Fi or LoRa) [11,24]. In this section, we present the primer knowledge of ambient signals including Bluetooth, ZigBee, TV, FM, LoRa, Wi-Fi, and LTE, so that readers can better understand the operation and development of Amb-BackComs systems.

2.3.1. Bluetooth [25]

Bluetooth is a short-range wireless technology standard, which is primarily designed for short-range, low-power, and low-cost wireless communication [26]. It consists of different versions among which Bluetooth 4.0, also known as Bluetooth Low Energy (BLE), is the most commonly encountered version.

BLE adopts Gaussian Frequency Shift Keying (GFSK) modulation with a bandwidth of 2 MHz [27]. In GFSK, bit '1' and bit '0' are represented by a positive and a negative frequency deviation, respectively. This means a single-tone signal can be generated by transmitting a stream of all zeros or ones. However, Bluetooth uses data whitening to avoid such sequences. Iyer et al. [28] have demonstrated how to generate a single-tone Bluetooth transmission and create a valid 802.11b Wi-Fi or ZigBee signal from this single-tone transmission. Most of the subsequent studies [26,29–31] on Bluetooth-based backscatter systems rely on the single-tone transmission since the single-tone signal provides a reliable carrier for backscatter. However, a fundamental limitation is that the single-tone transmission cannot be used for productive data communication, thereby greatly reducing spectrum efficiency.

2.3.2. ZigBee [32]

ZigBee builds upon the IEEE 802.15.4 standard [33]. The IEEE 802.15.4 standard specifies the physical and Medium Access Control (MAC) layers of ZigBee protocol stack while ZigBee defines the network and application layers. ZigBee radios adopt Direct-Sequences Spread Spectrum (DSSS) coding and Offset Quadrature Phase-Shift Keying (OQPSK) modulation. Such modulation creates the possibility for the tag to embed its data. Zhang et al. [22] have shown how a tag embeds its data in an OQPSK signal. To reduce signal interference, DSSS maps each ZigBee symbol into a corresponding pseudorandom noise (PN) code of 32 chirps. The chirps are then assigned into in-phase and quadrature series, and there exists a half-a-chip offset between in-phase and quadrature. Such offset also poses a challenge for phase-based tagdata modulation because the phase shift from the tag would damage this half-a-chip offset structure, thereby causing decoding failure on the receiver side. The challenge can be addressed by encoding one tag bit on multiple ZigBee symbols [34].

2.3.3. Television (TV) [35]

TV towers transmit continuous and uninterrupted signals. These signals are widely available in outdoor environments. Thus, TV signals are promising as a dependable source for backscatter. The studies of [1,36] have shown how to achieve ambient backscatter systems based on TV signals. While these systems are useful outdoors, they



Fig. 4. The structure of a baseband audio signal transmitted from a typical FM station [38,39].

are unsuitable for indoor environments. This is because these systems require the power level of TV signals to be at least -10 dBm to -20 dBm, but the strength of TV signals in indoor environments is very low and usually in the range from -80 to -50 dBm [37].

2.3.4. Frequency Modulation (FM)

FM stations broadcast continuous signals. These signals are ubiquitous in outdoor environments. They provide a source for ambient backscatter. An FM transmission can be mathematically expressed as follows [38].

$$FM(t) = \cos\left(2\pi f_c t + 2\pi\Delta f \int_0^t FM_{audio}(\tau) d\tau\right)$$
(6)

In Eq. (6), f_c , Δf , and $FM_{audio}(\tau)$ are the carrier frequency of the FM transmission, the deviation in frequency from f_c , and the baseband audio signal, respectively. The baseband audio signal consists of a mono audio stream, a stereo audio stream, a 19 kHz pilot tone, and a broad-cast data system (RDS) stream, as shown in Fig. 4. Wang et al. [38] have shown how to leverage FM transmissions for Amb-BackComs.

2.3.5. Long Range (LoRa) [40]

LoRa is a long-range and low-power wireless communication technology standard [41,42]. It adopts Chirp Spread Spectrum (CSS) modulation. In CSS, bits '0' and '1' are encoded respectively as a continuous chirp that increases linearly with frequency and a chirp that is cyclically shifted in time [11,43,44]. Fig. 5 shows a chirp '0' and chirp '1' in the time domain. As the chirp-modulated signal occupies the whole allocated bandwidth, it is less susceptible to channel noise, Doppler, and multi-path effects. Thus, it provides a reliable source for ambient backscatter. In addition, chirp-modulated signals can be decoded at very low signal-to-noise ratio (SNR). Thus, it is expected to enable longrange backscatter communication. LoRa uses packetized protocols to transmit data. A LoRa frame begins with a preamble. The preamble is composed of ten consecutive up-chirps with zero initial frequency offset. Guo et al. [45] have shown how to leverage the unique features in the LoRa preamble to detect the appearance of LoRa signals.

To demodulate the LoRa signal, the receiver first multiplies an incoming LoRa chirp with a down-chirp and then performs a Fast Fourier transform (FFT) on the resulting multiplication. This FFT operation will yield a peak in an FFT frequency bin. By tracking FFT peaks, the receiver finally demodulates the incoming LoRa symbol.

2.3.6. Wireless Fidelity (Wi-Fi) [46]

Wi-Fi usually refers to the wireless local area networking technique based on the IEEE 802.11 family of standards [47]. It is a shortrange wireless communication technique, which is primarily used for providing network communication capabilities to different devices in private, public, and commercial environments [48,49]. As most devices



Fig. 5. LoRa chirp signal '0' and chirp signal '1', in which BW is the LoRa chirp bandwidth [21].



Fig. 6. The 2.4 GHz Wi-Fi channels.

(e.g., Wi-Fi access points, smartphones, laptops, and watches) are Wi-Fi enabled, this makes Wi-Fi signals a natural choice for ambient backscatter and one of the most widely studied excitation signals in the literature. IEEE 802.11 supports various frequencies, such as 2.4 GHz and 5 GHz frequency bands. Regarding 2.4 GHz band, it is divided into 14 channels with a bandwidth of 22 MHz and spaced 5 MHz apart from each other except for a 12 MHz space before channel 14, as shown in Fig. 6. The channels labeled as 1, 6, and 11 are a set of non-overlapping channels. Thus, the backscatter tag in Wi-Fi backscatter systems usually uses a frequency phase of 20 MHz to avoid interference to the original signal.

There are many different versions of Wi-Fi, such as 802.11a [50], 802.11b [51], 802.11 g [52], and 802.11n [53]. Different versions of Wi-Fi use different techniques. In terms of 802.11n Wi-Fi, it supports Multiple-Input Multiple-Output (MIMO) technology. The technology can significantly improve the communication performance of wireless systems in terms of reliability and capacity [54]. Thus, if it is possible to acquire the benefits of the MIMO technology on low-power tags. Recent study [55] has implemented a MIMO backscatter system with a low bit error ratio and high throughput.

2.3.7. Long-Term Evolution (LTE) [56]

LTE radio transmits continuous signals. It adopts Orthogonal Frequency Division Multiplexing (OFDM) for downlink and Single Carrier-Frequency Division Multiple Access (SC-FDMA) for uplink transmission [57]. One LTE frame consists of ten 1 ms sub-frames. Each subframe consists of two 0.5 ms slots, and each slot consists of seven OFDM symbols. Each OFDM symbol consists of two main components: i) the useful symbol and ii) the cyclic prefix (CP). The useful symbol can be represented by multiplying the baseband signal with the carrier wave and the CP will be removed on the User Equipment (UE) side to avoid inter symbol interference (ISI). The LTE signal periodically embeds Primary Synchronization Signal (PSS) and Secondary Synchronization Signal (SSS) messages. These messages appear every five sub-frames and play a key role on the UE side. A key challenge in designing an LTE-based backscatter system is how to avoid changes to PSS and SSS messages. Chi et al. [58] have shown how to address this challenge and leverage LTE signals for Amb-BackComs.

3. Amb-BackComs systems

The Amb-BackComs paradigm first appeared in 2013, with the design of the first Amb-BackComs system, *Ambient backscatter* [1],



Fig. 7. Taxonomy of Amb-BackComs systems.

which allows two battery-free devices to communicate by using existing TV transmissions as the source of carrier and power. This eliminates the need for a new spectrum allocation, batteries, and a dedicated carrier source, thereby increasing spectrum efficiency and reducing the deployment as well as maintenance costs. In the last few years, Amb-BackComs systems have gained great attention within the research community owing to their remarkable advantages.

The available Amb-BackComs systems proposed in the literature support either one ambient signal or multiple ambient signals. Thus, these systems can be broadly categorized into two major groups, i) single-signal based Amb-BackComs systems and ii) multi-signal based Amb-BackComs systems. The systems that only work with a particular ambient signal are listed in the first group. The other group covers the systems that are compatible with multiple ambient signals. According to the type of ambient signals, the single-signal based Amb-BackComs systems can be further divided into the following six categories¹: i) Bluetooth based BackComs systems, ii) TV based BackComs systems, iii) FM based BackComs systems, iv) LoRa based BackComs systems, v) Wi-Fi based BackComs systems, and vi) LTE based BackComs systems. In addition, we primarily classify the multi-signal based Amb-BackComs systems based on the technique used. Considering that Wi-Fi is one of the most widely explored excitation signals in the area of Amb-BackComs, we further classify the Wi-Fi based BackComs systems based on the technique adopted. In the following sections, we review the existing literature based on the proposed taxonomy, which is schematically depicted in Fig. 7.

3.1. Single-signal based Amb-BackComs systems

3.1.1. Bluetooth based BackComs systems

Iyer et al. [28] introduce *Interscatter*, a BLE based backscatter system. The highlight of this system is that *Interscatter* tags can generate

valid Wi-Fi 802.11b or ZigBee signals by backscattering the incoming Bluetooth signals. As a result, a commodity Wi-Fi or ZigBee device can act as the backscatter receiver. The main idea of Interscatter is that by setting the appropriate bits on the payload part of the Bluetooth packet, a commodity Bluetooth device can transmit a single-tone signal (i.e., single-frequency), which can then be used to generate a valid Wi-Fi or ZigBee signal. Because Bluetooth adopts GFSK modulation, a single-tone signal can be created by transmitting a continuous stream of zeros or ones. However, a key challenge is how to create long sequences of either zeros or ones, since Bluetooth uses data whitening to avoid such sequences. The authors note that by setting bits in the payload section to be the same as the whitening sequence or their bit complement, we can generate long sequences of zeros or ones, respectively. To create Wi-Fi signals, the backscatter tag first shifts the signal from the Bluetooth channel to the Wi-Fi channel. However, existing sideband modulation techniques create the desired frequency shifts as well as the copy on the opposite side of the signal. This not only wastes bandwidth but causes interference. To address this, the authors design a singlesideband backscatter technique that only creates frequency shifts on a single side of the carrier. Then, the backscatter tag performs the modulation of DBPSK and DQPSK on the resulting signal-tone signal. By doing this, the backscatter tag can generate standards-compliant Wi-Fi packets on a single side of the resulting single-tone Bluetooth signal. The same procedure can also be applied to create ZigBee signals. Through experiments, the results show that Interscatter can achieve a range of around 90 ft when generating Wi-Fi signals, and a range of up to 15 ft when generating ZigBee signals.

Zhang et al. [29] present a novel BLE based backscatter system to enable multi-channel backscatter communication. In this system, a Bluetooth radio transmits BLE signals on one channel, while the backscatter tag reflects these BLE signals on multiple channels in a channel-hopping manner. To achieve this, the authors design a novel multi-channel backscatter tag. The key technique behind the tag is the *dynamic reconfiguration* technique. The dynamic reconfiguration provides multiple sets of different clocks. Each set of clocks can be used to shift BLE signals from one channel to another channel. Thus, multiple

¹ Existing ambient backscatter systems that support ZigBee radio also work with other ambient signals (e.g., Wi-Fi), and thus this group does not include ZigBee based backscatter systems.



Fig. 8. An example of direct frequency shift modulation [26].

sets of different clocks achieve multi-channel backscatter communication. The experimental results demonstrate that the proposed system can achieve a throughput of up to 2.8 Kbps.

Zhang et al. propose RBLE [26,30], a BLE backscatter system that tries to enable reliable backscatter communication. RBLE is built on the studies [22,28,29]. To do this, RBLE first presents a direct frequency shift modulation scheme that uses the single-tones part of BLE signals as the modulation carrier. The single-tone signal is generated by using the same technique as Interscatter [28]. Next, RBLE directly applies frequency shift to modulate symbols '0' or '1'. Specifically, suppose that the excitation signal is in the advertising channel 37, the single-tone part of this excitation signal is symbol '0', and the target channel is data channel 3. If we want to modulate a symbol '0' to the target channel, we only need to apply a frequency shift of 8 MHz. Otherwise, a frequency shift of 8.5 MHz is applied to modulate a symbol '1' into the target channel, as shown in Fig. 8. After that, RBLE introduces a dynamic channel configuration scheme. Similar to their previous work [29], the dynamic channel configuration scheme allows the backscatter tag to perform channel hopping, thereby reducing the interference to original signals and improving the reliability of backscatter communication. Then, RBLE presents a novel packet regeneration scheme. As such, the backscatter tag can regenerate an advertising packet or a data packet by modulating the incoming BLE advertising signal. This allows RBLE to communicate across all BLE channels, thereby further enhancing backscatter reliability. Finally, the authors evaluate the performance of RBLE in various real-world scenarios. The experimental results show that RBLE can achieve a throughput of 11.8 Kbps at 15 m.

Zhang et al. [31] test the Bit Error Ratio (BER) and Packet Error Ratio (PER) of RBLE [26] in an indoor scenario and find a huge performance gap between RBLE and commodity-level devices. In other words, RBLE cannot meet the BER and PER requirements (BER< 0.1% and PER< 17.5%) of the BLE specification [25,59]. Through analysis, the authors find that the backscatter reliability of *RBLE* is limited owing to its modulation scheme (i.e., BFSK modulation), since commodity BLE receivers are designed to receive and demodulate GFSK modulated BLE signals. Zhang et al. [31] thus design IBLE that tries to improve the reliability of BLE backscatter communication to the commodity level. The authors notice that the phase shift can be obtained by integrating the frequency over time. This means that digital frequency modulated signals can be transformed into digital phase-modulated signals. Based on this insight, two novel modulation schemes, i.e., Instantaneous Phase Shift (IPS) and GFSK modulations, are designed to improve the reliability of backscatter communication. Fig. 9 shows the architecture of IPS and GFSK modulations. At a high level, the IPS modulation can be viewed as the GFSK modulation without Gaussian Filter. On the IBLE tag side, the highlight is the design of a novel intermediate frequency (IF) signal synthesizer. This synthesizer leverages the output of the digital integrator module as the input and directly outputs a GFSK modulated IF signal or an IPS modulated IF signal. The generated IF signal will be used to modulate the excitation signal, thereby achieving BLE backscatter using IPS or GFSK modulation. Furthermore, a Forward Error Correction (FEC) coding is used to further enhance communication



Fig. 9. The architecture of IPS (the gray part) and GFSK (the whole part) modulations used in IBLE [31].

reliability. The experimental results show that IBLE can achieve a throughput of up to 8.275 Kbps and an effective uplink range of up to 20 m.

3.1.2. TV based BackComs systems

Liu et al. [1] design the first TV based BackComs system, in which two backscatter tags can communicate with each other by reflecting TV signals in the environment. In fact, this system is also the first Amb-BackComs system. In [1], the backscatter tag can serve as both a backscatter transmitter and a backscatter receiver. To do this, each tag is composed of three main components, i) a backscatter transmitter, ii) a backscatter receiver, and iii) a harvester. All components are connected to the same antenna and use the same ambient signals, as shown in Fig. 10(a). To transmit information, a switch is adopted at the transmitter, whose input is a stream of bits '0' and '1'. When the input is bit '0', the switch is in the non-reflecting state. Otherwise, if the input is bit '1', the switch is in the reflecting state. As such, a backscatter device can convey its data to another backscatter device, thereby allowing a backscatter tag to act as a backscatter transmitter. To extract backscattered information from ambient TV signals, a key insight is that the backscatter receiver can separate the backscattered signals and the ambient TV signals if the backscattered information is transmitted at a lower rate than that of ambient TV transmissions. This is possible because the difference in the bitrates of the two signals will result in adjacent samples in TV signals to be more uncorrelated than the adjacent samples in the backscattered signals. Thus, by averaging the received signals across multiple samples, the variations can be removed by the backscatter receiver in the ambient TV signals while remaining in the backscattered signals, thereby allowing the backscattered information to be decoded. Based on this insight, a low-power demodulator is designed on the receiver, which consists of an average envelope circuit, a compute-threshold circuit, and a comparator circuit, as shown in Fig. 10 (b). The received signals are first smoothed by the averaging circuit while yielding two voltage levels, V_0 and V_1 , corresponding to the power levels for zero and one bits, respectively. Then, a threshold value (i.e., $\frac{V_0+V_1}{2}$) is computed by the computethreshold circuit. Finally, the comparator circuit compares the average envelope signals with the threshold value to generate desired bits. The experimental results show that the backscatter receiver can achieve a bit rate of 10 Kbps over distances of less than 2 ft.

While the work of [1] achieves tag-to-tag communication by reflecting TV signals, its communication range and throughput are limited to 2 ft and 10 Kbps, respectively. This significantly limits its applicability. Parks et al. [36] thus propose another TV based backscatter system. In this system, a multi-antenna backscatter receiver (i.e., *u*mo) and a low-power coding mechanism (i.e., *u*code) are proposed to improve communication range and throughput. Through the experiments, the authors demonstrate that *u*mo increases the throughput from 10 Kbps to 1 Mbps and *u*code increases the communication range from 2 ft to up to 80 ft.



Fig. 10. (a) Block diagram of a backscatter device and (b) The demodulator [1].

3.1.3. FM based BackComs systems

Wang et al. propose *FM Backscatter* [38], the first FM based Back-Coms system. In this system, the backscatter tag can convey its data on top of ambient FM transmissions and any commodity FM receiver can act as the backscatter receiver. The highlight of this system is the design of the *overlay backscatter* modulation. It can transform the multiplicative nature of backscatter into an addition operation on the audio signals. Specifically, by picking the appropriate baseband signal B(t), such as $B(t) = cos \left(2\pi f_{back}t + 2\pi\Delta f \int_0^t FM_{back}(\tau) d\tau\right)$, in which f_{back} and $FM_{back}(\tau)$ are the center frequency of the baseband signal and the audio signal respectively, the backscattered signal $B(t) \times FM(t)$ can become:

$$\cos\left(2\pi f_{back}t + 2\pi\Delta f \int_0^t FM_{back}(\tau) d\tau\right) \times \cos\left(2\pi f_c t + 2\pi\Delta f \int_0^t FM_{audio}(\tau) d\tau\right)$$
(7)

Next, by applying the trigonometric identity, $2\cos(A) \times \cos(B) = \cos(A + B) + \cos(A - B)$, and using a single-sideband modulation technique [28], we can remove the term $\cos(A - B)$ and thus yield the Eq. (8), thereby achieving the conversion from a multiplication operation to an addition operation. In essence, Eq. (8) remains a valid FM signal, thereby allowing any FM receiver (e.g., a smartphone) to decode the tag information. The experimental results demonstrate that *FM backscatter* can achieve a throughput of up to 3.2 Kbps and a communication range of up to 60 ft.

$$\cos\left(2\pi\left[\left(f_{c}+f_{back}\right)+\Delta f\int_{0}^{t}\left(FM_{audio}\left(\tau\right)+FM_{back}\right)d\tau\right]\right)\tag{8}$$

Daskalakis et al. [60] propose a novel FM based backscatter system for smart agricultural monitoring, in which the backscatter tag can measure the moisture level in plants and transmit its data to a backscatter receiver by reflecting ambient FM broadcasting signals. To measure the moisture level in plants, an efficient solution is to measure the temperature difference between the leaf and the air [61]. To do this, the backscatter tag contains two analog temperature sensors, which measure the air temperature (T_{air}) and the leaf temperature (T_{leaf}), respectively. By subtracting T_{air} from T_{leaf} , we can obtain moisture level in plants. On the backscatter transmitter side, the backscatter tag adopts AM modulation and FM0 encoding on the FM transmissions to transmit its data. The experimental results show that the proposed system can achieve a bit rate of 500 bps and a communication range of 2 m.

Daskalakis et al. [39] extend the work of [60]. In [39], the backscatter tag can read four analog inputs by using a 16-channel 12-bit analog-to-digital converter (ADC), thereby enabling a wide range of applications. In addition, a novel *receiver algorithm* is designed at the backscatter receiver to improve throughput and communication range. The key idea of this algorithm is to reduce the frequency difference between the backscatter tag and the backscatter receiver. A matched filter is then employed to reduce noise and interference and maximize the SNR of the received signals, thereby improving the throughput and communication range of the system. The experimental results show that the proposed system can achieve a bit rate of 2.5 Kbps at a tag-to-reader distance of 5 m.

Vougioukas et al. [62] present a novel design that can achieve a Tag-to-FM receiver distance of 26 m and 15 m, outdoors and indoors, respectively, when reflecting FM transmissions, and work with any off-the-shelf FM receiver (e.g., a smartphone). To achieve this, two pivotal techniques, i.e., FM Remodulation and Selection Diversity, are proposed. The FM remodulation technique uses the same ideas as the overlay backscatter technique used in [38]. Specifically, the tag signal is designed with the same structure as the FM radio transmissions. Thus, the reflected signal is still a valid FM signal, thereby allowing any off-the-shelf FM receiver to decode the tag data. Selection diversity leverages the fact that modulation from the tag signal occurs at passband, not at baseband, and thus the tag signal can be modulated on multiple FM radio signals at the same time. By selecting the FM station with the strongest received power, we can increase the value of the amplitude of the backscattered signal, which in turn reduces the impact of thermal noise at the receiver's output, thereby achieving a longer communication range from the tag to the receiver.

Another FM based backscatter system is introduced by Vougioukas et al. in [63]. In this system, the backscatter tag can use both analog and digital modulation schemes. As a result, the system can benefit from different modulations. In terms of the digital modulation scheme, the backscatter tag uses Pseudo-Frequency Shift Keying (P-FSK) [64], a frequency-shifted form of On-Off Keying (OOK), and frequency-shifted Binary Phase Shift Keying (S-BPSK), a frequency-shifted form of Binary Phase Shift Keying (BPSK). Unlike P-FSK, the S-BPSK does not require information about the ambient carrier. In addition, the short packet error correction coding can be utilized for the S-BPSK. The analog modulation scheme is designed based on *FM Remodulation* principles [62]. The experimental results show that the proposed system can achieve a tag-to-receiver distance of 26 m in an outdoor environment while consuming the power of 24 μW .

3.1.4. LoRa based BackComs systems

Peng et al. propose *PLoRa* [21], a novel LoRa based backscatter system that tries to enable long-range backscatter communication. In this system, the tag can transmit its data by backscattering incoming LoRa signals and the backscattered signal remains a valid LoRa signal. On the backscatter tag side, to implement *PLoRa*, a challenge is how to detect an incoming LoRa packet and synchronize with its LoRa symbols on the low-power tag without requiring power-hungry components (e.g., *digital to analog converter* (DAC) and *voltage-controlled oscillator* (VCO)). Peng et al. [21] note that the power consumption of packet detection decreases monotonically with the increase of sampling rate [65], and thus propose to reduce the sampling rate for packet detection. Regarding synchronization, the *PLoRa* tag first performs the correlation between incoming signals and pre-stored preambles. After detecting the whole preamble of the incoming signal and waiting for a specific time, the *PLoRa* tag begins to backscatter. Another challenge is

how to generate a standard LoRa signal in the backscattering process. To address this, a novel blind chirp modulation scheme is designed. The main idea underlying scheme is that the incoming LoRa chirp can be shifted by an amount of $\frac{BW}{2}$ and $\frac{-BW}{2}$, in which BW denotes the LoRa chirp bandwidth, by multiplying an FSK-modulated baseband signal with the incoming LoRa chirp, and these two shifted LoRa chips can be spliced together into a valid LoRa chirp. To reduce interference, moreover, PLoRa shifts the backscattered signal into a different channel. On the backscatter receiver side, a novel backscatter signal demodulation scheme is proposed to demodulate the tag data. The main principle of this scheme is to examine the location consistency of FFT peaks. Specifically, the backscatter receiver will perform two FFT operations, one on the product of the excitation LoRa chirp and the backscattered LoRa chirp, and the other on the product of the excitation LoRa chirp and the LoRa down chirp. These two FFT operations will lead to two peaks in FFT bins. If the location difference of the two peaks is greater than a pre-defined threshold, this means a bit '1' is transmitted. Otherwise, a bit '0' is transmitted. Moreover, an energy management circuit is designed to efficiently manage node energy. Through the experiments, the authors demonstrate that PLoRa can achieve a communication range of up to 1.1 km and a bit rate of 6.25 Kbps at 300 m.

Another LoRa based BackComs system, Aloba, is introduced by Guo et al. in [45,66]. In Aloba, the tag can detect LoRa transmissions even if there exist interfering signals (e.g., noise and other legacy signals). Guo et al. notice that the LoRa preamble consists of ten consecutive up-chirps with zero initial frequency offset. When the incoming signal passes through a low-pass filter, the LoRa preamble will show ten equally spaced RSS pulses, while the interfering signals will not. Therefore, Aloba can detect LoRa transmissions by leveraging this unique feature. After detecting the LoRa preamble as well as waiting for a specific time, the Aloba tag begins to perform modulation using OOK. Unlike PLoRa [21], the backscattered signal in Aloba works in the same frequency band of the carrier signal. On the backscatter receiver side, hence, the received signals are the superposition of the backscattered signal and the carrier signal. To decode the tag data, Aloba first converts the received LoRa signals into a sinusoidal tone. As a result, the variation of the sinusoidal tone is affected by the backscatter signal. Then, Aloba leverages the variations to demodulate the tag data. The experimental results show that Aloba can achieve a throughput of 199.4 Kbps at 50 m.

3.1.5. Wi-Fi based BackComs systems

Most of the existing Wi-Fi based BackComs systems leverage the principle of *codeword translation*, while others use channel information, MAC-layer features, or optimization code. In the following, we provide a brief overview of these systems. We primarily classify them based on the technique used.

(1) Channel information based: Kellogg et al. [67] design Wi-Fi Backscatter, which is the first Wi-Fi based BackComs system that enables the communication between backscatter tags and off-the-shelf Wi-Fi devices. This system supports both uplink and downlink communications. In [67], the backscatter tag transmits bit '1' or bit '0' by reflecting or absorbing the Wi-Fi packets, respectively. As a result, the channel information, such as received signal strength indicator (RSSI) and channel state information (CSI), will be changed on the backscatter receiver side. Thus, by measuring the changes in RSSI or CSI, the backscatter receiver can decode the tag data. The experimental results demonstrate that the proposed system can achieve a data rate of up to 1 Kbps and a range of up to 2.1 m.

(2) Codeword translation based: The codeword translation technique is first introduced in *HitchHike* [68]. This system can work completely with commodity 802.11b Wi-Fi devices thanks to the technique. The authors [68] note that every 802.11b Wi-Fi packet can be viewed as a sequence of codewords from the same codebook. Thus, the backscatter tag can embed its data by transforming the received codeword into another valid codeword in the 802.11b codebook. In this way, the backscattered packet remains a valid 802.11b Wi-Fi packet and thus can be decoded by commodity 802.11b devices. To decode the tag information, *HitchHike* uses two receivers, one for receiving Wi-Fi signals being transmitted and the other for backscattered signals. The tag data can be extracted by XORing the data from the two receivers. To avoid self-interference, moreover, *HitchHike* shifts the backscattered signal into a non-overlapping Wi-Fi channel. The experimental results show that *HitchHike* can achieve an uplink throughput of up to 300 Kbps at 34 m in a line-of-sight scenario and an uplink throughput of around 100 Kbps at 25 m in a non-line-of-sight scenario.

Zhao et al. introduce *MOXcatter* [20], a novel Wi-Fi based BackComs system that tries to enable spatial stream backscatter using commodity multi-antenna Wi-Fi. *MOXcatter* is built on the work of *HitchHike* [68]. In *MOXcatter*, the backscatter tag uses phase change (similar to the *codeword translation* method in *HitchHike* [68]) to embed its data on incoming Wi-Fi signals. In particular, when backscattering 802.11n signal-stream signals, the backscatter tag applies a phase shift (e.g., 0 or 180 degrees) to symbols in a Wi-Fi packet to embed desired data. When backscattering 802.11n multi-stream signals, the backscatter tag applies a phase shift to the whole data payload of a packet. In other words, the antennas on the tag transmit the same data. To decode the tag data, like *HitchHike* [68], *MOXcatter* also requires two receivers. The experimental results demonstrate that *MOXcatter* can reach a bitrate of up to 50 Kbps for 802.11n signal-stream signals and up to 1 Kbps for 802.11n double-stream signals with a range up to 14 m.

Wang et al. design RapidRider [69], a Wi-Fi based BackComs system that enables single-symbol modulation. This system is built on the studies of HitchHike [68] and MOXcatter [68]. Unlike HitchHike and MOXcatter, RapidRider modulates one bit in each symbol and requires one receiver only. The authors note that by using a reference symbol, both productive and tag data can be combined into the same packet. Thus, one receiver is able to decode the tag data. To enable singlesymbol modulation, Wang et al. [69] first analyze the reasons of the failure of the single-symbol modulation in existing Wi-Fi based backscatter systems and find that the root cause is the incompatibility between symbol-wide and payload-wide operations. Based on this insight, RapidRider performs the decoding on the deinterleaved-data level instead of the payload level. On the backscatter receiver side, a novel deinterleaving-twins decoding scheme is proposed, which leverages backward deinterleaved data and forward deinterleaved data to recover the tag data. The experimental results show that RapidRider can achieve a throughput of 237.8 Kbps with ambient OFDM signals.

Zhao et al. [70] present a multi-hop Wi-Fi based backscatter system, namely X-Tandem. In this system, the backscatter tag can not only work as relays for each other but also modulate their data simultaneously in a single backscatter packet. On the backscatter tag, the employed modulation is similar to the codeword translation method in [68]. Thus, the backscatter packet remains a valid Wi-Fi packet that can be decoded by commodity Wi-Fi devices. The main principle of X-Tandem is shown in Fig. 11(a). Assume there is an ongoing Wi-Fi signal, the first tag uses this original Wi-Fi signal as an excitation, modulates its data onto a specific part of this excitation (see Fig. 11(b)), and shifts backscattered signal into a new channel to reduce interference. The second tag, when receiving the original packet, performs the same procedure. As a result, it embeds its data in the same backscattered packet and relays the first tag's data. The experimental results show that, for a two-hop implementation, X-Tandem can achieve a throughput of up to 200 bps when the distance between the two tags is 0.05 m and a communication range of up to 8 m.

Liu et al. [55] design and implement another Wi-Fi MIMO backscatter system, namely VMscatter. Unlike MOXcatter [20], the antennas on the VMscatter tag transmit different data. In addition, VMscatter can achieve the same full diversity gain as the traditional MIMO system and work with massive MIMO. The diversity features of MIMO can be leveraged in VMscatter by operating the space-time coding scheme.



Fig. 11. (a) X-Tandem communication and (b) The data field allocation in the original packet for multiple backscatter tags [70].

A highlight of this system is that the *VMscatter* tag implements the space–time coding by using a *two-states* (on and off) switch since the space–time coding encodes data into four states. Moreover, to successfully demodulate the tag data, *VMscatter* deals with the frequency mismatch between the tag and the receiver, and evaluates the channels among the sender, tag, and receiver. The experimental results show that *VMscatter* can achieve a bit error rate of around 0.000011 and the data rates of 239.7 Kbps and 245.9 Kbps for single data stream and multi-data stream, respectively.

Dunna et al. [71] notice that many existing studies in the area of Wi-Fi based BackComs overlook the importance of synchronization, thereby leading to limited downlink range and reduced throughput. Dunna et al. [71] thus present SyncScatter that tries to achieve synchronized backscatter communication with incident Wi-Fi signals. To implement SyncScatter, the authors first analyze the synchronization requirement for Wi-Fi based BackComs and the sensitivity requirement needed at the tag, and conclude that the sensitivity of -35 dBm would provide the maximum range benefit and the minimum required synchronization accuracy is 150 nanoseconds for 802.11b Wi-Fi signals. However, achieving these requirements typically consumes significant power, which is not feasible for the passive tag. To address this problem, a novel hierarchical wake-up scheme is designed at the tag. The key idea behind this scheme is to set wake-up functionality apart from synchronization functionality. Specifically, a low-power wake-up receiver is used to detect the appearance of the excitation signal and wake up the synchronization receiver. After that, the synchronization receiver leverages high-power radio frequency amplification to provide the required synchronization accuracy and desired sensitivity. By periodically turning the amplification on and off, the power consumption of the overall system can be reduced significantly. The experimental results demonstrate that SyncScatter can achieve a communication range of up to 30 m and a bit rate of 500 Kbps, with an average power consumption of 30 μW .

Liu et al. [72] note that existing OFDM-based Wi-Fi backscatter systems [20,68,70] leverage the principle of *codeword translation* at the OFDM symbol-level to send the tag data. To make codeword translation work, these systems require specific Wi-Fi receivers that cannot enable the phase error correction. This is because the phase error correction in Wi-Fi receivers can eliminate the phase offset created by a tag, which results in the decoding error. Liu et al. [72] thus propose TScatter. The highlight of this system is the design of a sample-level modulation scheme. In this scheme, the phase offsets on the pilot subcarriers and the data subcarriers are very different. Therefore, the phase offsets on the data subcarriers cannot be eliminated by the phase error correction, since the phase error correction leverages pilot subcarriers to correct the phase offset. On the backscatter receiver side, a novel demodulation model is proposed and the tag data is estimated by using a minimization function that is derived from the proposed demodulation model. Liu et al. [72] evaluate the performance of TScatter in various real-world scenarios. The results show that, with 4-phase modulation scheme, TScatter can achieve a throughput of around 10.61 Mbps over a range of 120 ft in a hallway scenario.

(3) Other approaches: Abedi et al. propose WiTAG [73,74], which leverages MAC-layer features to enable communication between backscatter tags and Wi-Fi devices. Specifically, the IEEE 802.11n/ac standards support a MAC-layer frame aggregation mechanism to reduce communication overhead. In this mechanism, multiple MAC Protocol DATA Units (MPDUs) are integrated into a larger aggregated frame (A-MPDU). After receiving the A-MPDU, the receiver will transmit a block acknowledgment (ACK) back to the sender to report the state of each MPDU in the A-MPDU. Therefore, during the transmission of each MPDU, a WiTAG tag can transmit bits '0' or '1' by corrupting (e.g., changing the phase of a signal) or not corrupting the transmission, respectively. The experimental results show that, with a distance of 8 m between the sender and the receiver, WiTAG can achieve a throughput of 40 Kbps when the tag operates anywhere between them.

He et al. [75] note that most Wi-Fi based backscatter systems suffer from low reliability owing to the intermittent characteristic of the Wi-Fi signals. Thus, a novel Wi-Fi based BackComs system, namely GuardRider, is proposed, which tries to achieve reliable communication between backscatter tags and backscatter receivers. A highlight of this system is the design of an optimized Reed-Solomon (RS) code scheme. The main principle of this scheme is to adaptively add some redundancy in a given information sequence, so that the tag data and redundant bits will be respectively transmitted during the "on" (i.e., presence) and "off" (i.e., absence) states of Wi-Fi signals. To implement GuardRider, a key problem is to evaluate the "on" and "off" states of Wi-Fi signals. Considering the limited resources of the tag, the authors shift the estimating process from the tag to the receiver. On the backscatter receiver side, the receiver first listens on the desired channel. After that, the duration of the "on" and "off" states of Wi-Fi signals is measured based on the received signals. Then, the optimal RS code parameters are obtained based on both the duration and a novel heuristic algorithm. Finally, the receiver feeds back the parameters to the tag. Through simulations and experiments, the results show that GuardRider can achieve a throughput of around 700 Kbps.

3.1.6. LTE based BackComs systems

Chi et al. [58] design the first LTE based BackComs system called LScatter. It enables ubiquitous, high throughput, and low latency Back-Coms by reflecting the continuous and ubiquitous LTE signals. To achieve this, on the backscatter tag, a challenge is to avoid modification of the primary synchronization signal (PSS) and secondary synchronization signal (SSS) messages, because these messages are important for decoding the LTE signal. The authors thus design an ambient LTE signal synchronization circuit to detect the PSS signals. The basic idea underlying the circuit is that the signal strength of PSS in LTE signal is higher than that of other parts in LTE signal. Therefore, by comparing the average envelope signals with the incoming envelope signals, we can determine whether the PSS appears. Once the PSS signals are detected, the backscatter tag can avoid modulating the PSS and SSS by transmitting square waves. Another challenge is how to modulate the LTE signal to achieve higher throughput. The authors note that the LTE signal in time-domain is constructed by the basic-timing units. The basic-timing units provide a much higher granularity than a symbol duration. The authors thus design a novel phase modulation scheme. It embeds tag data by changing the phases of the LTE signals in different basic-timing units, thereby achieving high throughput. On the backscatter receiver side, the reference signals on different subcarriers in the original LTE physical layer are used to eliminate the effect from the phase offset. The experimental results demonstrate that *LScatter* can achieve a throughput of up to 13.63 Mbps.

Discussion: In this section, we have reviewed single-signal based BackComs systems. Table 2 provides a summary of different Amb-BackComs systems from the performance perspective, including minimum power, maximum throughput, and maximum range. From the table, we observe that *LScatter* [58] achieves a maximum throughput. However, it is not compatible with existing infrastructures owing to the need for a dedicated receiver. This limits its application in the real world. In general, Bluetooth, ZigBee, and Wi-Fi signals would be a more appropriate choice for indoor applications while outdoor applications can be implemented by leveraging TV, FM, LoRa, and LTE signals. In terms of single-signal based BackComs systems, some open issues are summarized as follows:

- As Amb-BackComs leverage signals from existing systems, it is important to ensure that backscattering will not cause any performance degradation to existing systems. Therefore, the evaluation and analysis of the interference to existing systems is a hot research direction.
- Many current studies in this area mainly focus on uplink communication. Performance improvement in downlink communication is less studied. Some techniques (e.g., full-duplex) can be exploited to improve the overall throughput.
- The existing studies in this area mainly consider the communication between one tag and one receiver. However, practical applications involve multiple tags. Thus, the coordination between multiple tags needs to be further studied.
- In terms of Bluetooth based BackComs systems, many existing studies are built on the work of *Interscatter* [28]. These studies leverage a single-tone transmission for backscatter. As a result, the excitation signals cannot be used for productive data communication and carriers simultaneously, thereby greatly reducing spectral efficiency. Thus, more studies need to be done to improve spectral efficiency.
- For Wi-Fi based BackComs systems, many current studies in this area leverage the principle of *codeword translation*. While *codeword translation* works well in single-stream excitation, it has a low throughput in multi-stream excitation. Thus, new modulation schemes are needed to be studied.

3.2. Multi-signal based Amb-BackComs systems

We primarily classify the multi-signal based Amb-BackComs systems based on the technique used.

(1) Frequency shifting based: The frequency shifting technique is first introduced in FS-Backscatter [76]. In this system, the backscatter tag is able to shift an incoming Bluetooth or Wi-Fi signal into an adjacent Bluetooth or Wi-Fi band. This leads to a reduction in interference between the backscattered signal and the original signal. As such, the tag can transmit its information by using on-off keying (OOK) modulation. Shifting the carrier by 20 MHz, however, typically needs to consume milliwatts of power, which is not feasible for low-power tags. To address this, a novel ring oscillator is designed on the backscatter tag, which can consume tens of micro-watts while shifting the carrier by such a substantial amount. The experimental results show that FS-Backscatter can achieve a throughput of 627.7 bps and an operational distance of up to 4.8 m when reflecting Wi-Fi signals, and a throughput of up to 50 Kbps at 36 m when reflecting BLE signals.

(2) Codeword translation based: Zhang et al. [22] notice that any wireless signal on the ISM band can be viewed as a sequence of codewords from a codebook to represent different bits that are being transmitted. This means that the codeword translation technique originated in [68] can be extended to support other various commodity radios, such as Bluetooth and ZigBee. Zhang et al. [22] thus propose FreeRider, a multi-signal based BackComs system that works with Bluetooth, 802.11g/n Wi-Fi, and ZigBee radios. In this system, the excitation signal can carry productive data and serve as the carrier at the same time. In addition, this system requires two receivers to decode the tag information. The authors evaluate the performance of FreeRider in various scenarios. The experimental results show that FreeRider can achieve a data rate of around 60 Kbps at 18 m when reflecting 802.11g/n Wi-Fi signals, a throughput of 12 Kbps at a distance of 20 m when reflecting ZigBee signals, and a data rate of around 50 Kbps at 10 m when reflecting Bluetooth signals.

Gong et al. propose Multiscatter [34], a novel multi-signal based BackComs system. Like FreeRider [22], Multiscatter also supports multiple radios (e.g., ZigBee, Bluetooth, and Wi-Fi) and these radios are simultaneously used for productive data communication and carriers. Different from FreeRider, Multiscatter tags can distinguish those excitation signals and leverage excitation diversity to improve throughput gains. In addition, Multiscatter uses a single receiver to decode the tag data. To achieve the goal of identifying different signals, a key insight is that every packet has a unique preamble field. Thus, by identifying such preambles, different signals can be distinguished. Doing so first requires obtaining high-bandwidth baseband signals, like ZigBee. However, acquiring such signals usually requires power-hungry components, which are infeasible for low-power tags. The authors thus propose a highbandwidth rectifier, which can acquire high-bandwidth input signals by using simple low-power hardware elements, such as diodes, capacitors, and resistors. In addition, a low-power protocol identification scheme is designed to further reduce computation and power overhead while keeping identification results accurate. After identifying the excitation signals, the tag embeds its data on those signals by using a novel overlay modulation scheme. The overlay modulation is based on codeword translation [22]. The highlight of overlay modulation is using the reference symbol in the carrier. The reference symbol plays the same role as the original Wi-Fi signal used in the decoding process of the codeword translation. As a result, a backscatter receiver is able to complete decoding. The experimental results show that Multiscatter can achieve maximum backscatter communication ranges of 28 m, 22 m, and 20 m for Bluetooth, 802.11b/n Wi-Fi, and ZigBee, respectively.

(3) Other approaches: Jung et al. [77] propose GateScatter, a multisignal based BackComs system that works with Bluetooth and ZigBee signals. In this system, the GateScatter tag is able to transform an incoming ZigBee or Bluetooth signal into a valid 802.11b Wi-Fi signal while maintaining the original data. In short, the GateScatter tag can be viewed as a gateway. To implement GateScatter, the authors note that, with an appropriate tag signal, the incoming ZigBee's positive (negative) half-sine signal after backscattering approximates 802.11b Wi-Fi's non-inverted (inverted) Barker code. This means a ZigBee signal can be reshaped into an 802.11b Wi-Fi signal on the backscatter tag. However, the tag signal has to be carefully selected because both in-phase and quadrature signals of ZigBee can correlate to 802.11b Wi-Fi Barker code, which results in false decoding. The authors notice that the original ZigBee symbols can be uniquely identified with only the quadrature portion of ZigBee. Hence, an optimization method is proposed to select the optimal tag signal, so that only the quadrature component correlates to the Wi-Fi Backer code. On the receiver side, after the received signal has been decoded into Wi-Fi bits, a mapping from Wi-Fi bits to ZigBee symbols is performed to retrieve the original ZigBee symbols. Furthermore, the authors demonstrate that GateScatter works with BLE. The experimental results show that GateScatter can achieve a data rate of up to 662 Kbps and a communication distance at most 27 m when backscattering Bluetooth signals, and a throughput

Table 2 Summary of Amb-BackComs systems.

Article	Year	Signal source	Performance				
			Minimum Power	Maximum Throughput		Range	
				Throughput	Distance	TX to Tag	Tag to RX
Interscatter [28]	2016	BLE	28 μW (IC)	11 Mbps	N/A	0.9 m	27.4 m
[29]	2020	BLE	N/A	2.8 Kbps	1 m	0.5 m	23 m
RBLE [26]	2020	BLE	37 μW (IC)	16.6 Kbps	1 m	N/A	56 m
IBLE [31]	2021	BLE	53.528 mW	8.275 Kbps	2 m	0.5 m	20 m
Ambient backscatter [1]	2013	TV	0.79 μW	10 Kbps	0.4 m	N/A	2.5 m
[36]	2014	TV	430.9 μW	1 Mbps	2.1 m	N/A	N/A
FM backscatter [38]	2017	FM	11.07 μW (IC)	3.2 Kbps	4.9 m	N/A	18.3 m
[60]	2017	FM	N/A	500 bps	0.5 m	N/A	2 m
[39]	2017	FM	0.677 mW	2.5 Kbps	5 m	34 km	5 m
[62]	2017	FM	24 μW	N/A	N/A	N/A	26 m
[63]	2018	FM	24 µW	N/A	N/A	N/A	26 m
PLoRa [21]	2018	LoRa	2.591 mW	6.25 Kbps	300 m	N/A	1.1 km
Aloba [45]	2020	LoRa	0.3 mW	199.4 Kbps	50 m	1 m	250 m
Wi-Fi backscatter [67]	2014	Wi-Fi	9.65 μW	1 Kbps	N/A	N/A	2.1 m
HitchHike [68]	2016	802.11b Wi-Fi	33µW (IC)	300 Kbps	34 m	1 m	54 m
MOXcatter [20]	2018	802.11n Wi-Fi	N/A	50 Kbps	3 m	0.3 m	14 m
RapidRider [69]	2021	802.11a/g/n Wi-Fi	N/A	239.1 Kbps	1 m	N/A	16 m
X-Tandem [70]	2018	802.11b/g/n Wi-Fi	14.2 mW	200 bps	0.5 m	0.3 m	8 m
VMscatter [55]	2020	802.11n Wi-Fi	32 μW (IC)	500 Kbps	1.5 m	0.9 m	34.6 m
SyncScatter [71]	2021	802.11b Wi-Fi	30 µW (IC)	N/A	N/A	N/A	30 m
TScatter [72]	2021	802.11 g Wi-Fi	30.2 µW (IC)	10.61 Mbps	N/A	N/A	48.8 m
WiTAG [73]	2018	802.11n/ac Wi-Fi	N/A	40 Kbps	7 m	1 m	17 m
GuardRider [75]	2019	Wi-Fi	N/A	700 Kbps	N/A	N/A	N/A
LScatter [58]	2021	LTE	153 μW (IC)	13.63 Mbps	N/A	0.6 m	97.5 m
FS-backscatter [76]	2016	Wi-Fi	45 μW	627.7 bps	N/A	N/A	4.8 m
		BLE		50 Kbps	3.6 m	N/A	4.4 m
		802.11g/n Wi-Fi		60 Kbps	18 m	1 m	42 m
FreeRider [22]	2017	Bluetooth	32 μW (IC)	50 Kbps	10 m	1 m	12 m
		ZigBee		14 Kbps	12 m	1 m	22 m
		802.11b/n Wi-Fi		219.8 Kbps	2 m	0.8 m	28 m
Multiscatter [34]	2020	BLE	279.5 mW	278.4 Kbps	2 m	0.8 m	22 m
		ZigBee		26.2 Kbps	2 m	0.8 m	20 m
GateScatter [77]	2020	BLE	77.4 µW (IC)	662 Kbps	1 m	0.5 m	23 m
		ZigBee		222 Kbps	1 m	0.5 m	27 m
		Wi-Fi		10 Kbps	3.5 m	N/A	3.5 m
Glaze [78]	2019	FM	850 μW	10 Kbps	3.5 m	N/A	10 m
		TV		10 Kbps	3.5 m	N/A	9 m
[70]	2017	FM	N/A	1 Kbps	15.3 m	N/A	22 m
[/9]		TV		1 Kbps	9 m	N/A	12.4 m

Note: Under "Range", we provide the maximum operating distance between tag and receiver (denoted by *Tag to RX*) and the corresponding distance between ambient source and tag (denoted by *Tx to Tag*). In the "Maximum Bit Rate" column, the "Distance" is the distance between tag and receiver when obtaining the maximum throughput. For all numbers marked "IC" in the "Minimum Power" column, the number is the simulation result of an integrated circuit (IC) design.

of up to 222 Kbps and a communication range of up to 27 m when backscattering ZigBee signals.

Kapetanovic et al. [78] propose Glaze, which tries to overlay additional data on existing ambient signals (e.g., Wi-Fi and Bluetooth) to enable downlink communication while those signals are still being received by legacy receivers. Kapetanovic et al. [78] notice that most wireless receivers are able to tolerate a small amount of noise in the wireless signal. Thus, we can transmit additional data on existing ambient signals by introducing small amounts of attenuation. Based on this insight, a novel Glaze module is designed, which can overlay data on existing signals by changing the amplitude of those signals. The Glaze module is connected to the antenna ports of existing wireless transmitters, as shown in Fig. 12. The input of the Glaze module is a wireless signal and the output is a composite signal that is composed of original information and additional data. On the Glaze receiver side, the composite signal passes through the matching network circuit, envelop detector circuit, threshold compute circuit, and comparator circuit, and finally is decoded. The experimental results demonstrate that Glaze can achieve downlink ranges of 3.5 m, 10 m, and 9 m by overlaying data on Wi-Fi, FM, and TV signals, respectively.

Another multi-signal based BackComs system is introduced by Yang et al. in [79] to enable the backscatter system to operate on ambient carriers as low as -80 dBm. A highlight of this system is the design of an ultra-wideband backscatter tag. This tag can reflect any ambient signal from 80 MHz to 900 MHz, including TV, FM, and cellular



Fig. 12. The Glaze module connects to the antenna port of wireless transmitters [78].

signals, to transmit its information. By combining multiple backscatter carriers, the signal-to-noise ratio can be enhanced, thereby improving the sensitivity of a backscatter receiver. To decode the backscattered data, a novel signal processing module is designed, as shown in Fig. 13. First, on the receiver side, the received signals are digitally sampled by an analog-to-digital converter. Then, each signal from different channels is separated using an appropriate band pass filter. After that, the backscattered signals embedded in different channels are individually extracted by down-conversion and signal processing operations. Finally, a *maximal ratio combining with interference* technique is used to decode the backscattered data. The experiment results show that the proposed system can achieve a communication range of up to 22 m at



Fig. 13. The block diagram of the receiver signal chain used in [79].

data rates of 158 bps when reflecting FM signals, and a distance of 9 m at 1 Kbps when reflecting TV signals.

Discussion: In this section, multi-signal based Amb-BackComs systems are reviewed and summarized in Table 2. Compared to singlesignal based Amb-BackComs systems, multi-signal based Amb-BackComs systems have advantages in scenarios where a particular type of ambient signals is not available or the signal quality is poor. However, the design of multi-signal based Amb-BackComs systems is more complicated and the performance is limited in terms of the communication range and throughput. Thus, we should choose Amb-BackComs systems based on the actual requirements. Similar to the issues in single-signal based Amb-BackComs systems, some critical problems (e.g., multiple access, downlink communication, and interference to existing systems) are less studied in multi-signal based Amb-BackComs systems. Therefore, there is a need for further studies to address these problems.

4. Applications

Ambient backscatter technology is expected to be widely adopted in various practical applications (e.g., smart homes [20,70], environment monitoring [60], and health care [28]) by virtue of its characteristics of ultra-low-power, efficient spectrum utilization, low cost and easy deployment. In this section, we provide a brief discussion on the practical applications empowered by Amb-BackComs.

4.1. Smart homes and smart cities

In a smart home, massive passive tags can be flexibly deployed at any location, such as embedded in walls and furniture. These devices are powered by collecting energy from ambient signals (e.g., Wi-Fi, Bluetooth) and can collaborate to monitor the indoor environment. The environmental data (e.g., temperature, humidity) collected by the tag can be reported to smart home devices. By analyzing those data, smart home devices can intelligently adjust the temperature or humidity of the room, thereby providing a comfortable living condition for us. The studies in [20,70] have demonstrated the applicability of smart home design using Wi-Fi backscatter.

Similarly, in a smart city, massive passive tags can be deployed in buildings, streets, and parking spaces to help improve the quality of life by providing users with helpful information (e.g., parking availability). Wang et al. [38] have demonstrated that smart posters are able to convey information (e.g., audio and notifications) to nearby users by backscattering FM radio signals. Waghmare et al. [80] have shown that posters can be used to collect user feedback (e.g., shopping experience) and transmit the collected information to a nearby smartphone using FM backscatter.

4.2. Universal localization

Localization is a process that finds a device or user location in any arbitrary environment such as a home or office. Obtaining the location of devices or users is becoming a fundamental requirement in our daily life. A person may desire to easily obtain the location of his or her items such as keys, smartphone, pill bottle, or books. By attaching a passive tag to these objects of interest, we can find them by leveraging the ambient backscatter communication technique. *WiTag* [81] can achieve a median localization error of 0.92 m in a line-of-sight scenario and 1.48 m in a non-line-of-sight scenario, in an office building with Wi-Fi access points. *TagFi* [82] can achieve a median localization accuracy of 0.2 m over a range of up to 8 m in an office building with Wi-Fi infrastructure.

4.3. Environmental monitoring

Environmental monitoring is increasingly important for tackling global climate change and improving the environment. Ambient backscatter communication promises to play a key role in environmental monitoring owing to low deployment and maintenance costs. Backscatter tags can be placed in any arbitrary environment to assist in monitoring environmental parameters such as humidity and temperature. Temperature and humidity data measured by the tag can be collected and further analyzed so that a corresponding plan is implemented. Daskalakis et al. [60] have shown an ambient FM backscattering scheme for smart agricultural monitoring.

4.4. Health care

Health care is vitally important for preventing or treating illness and injuries. Traditional health care is carried out in hospitals or clinics, with the help of medical apparatus and instruments. It is inconvenient for the elderly and office workers, since they may desire to diagnose their health even without going to the hospital. Ambient backscatter communication promises to achieve this vision. In particular, the backscatter tags can combine a variety of biosensors and collect physiological signs such as body temperature and heart rate. By backscattering ambient signals (e.g., Wi-Fi), these tags can report the collected physiological information to a receiver (e.g., a smartphone). In this way, users can easily know their health conditions. In addition, these physiological data can be transmitted to their family members, so that families can take better care of them. Iyer et al. [28] have demonstrated that a smart contact lens can leverage ambient backscatter communication to help diabetic patients sense their condition.

4.5. Smart card

Owing to the small size, backscatter tags can be embedded inside the cards. In this way, these cards are able to communicate with each other by leveraging ambient backscatter technology. A typical application scenario is the money transfer between smart cards. In [1], the authors present a proof-of-concept of the smart card application. Specifically, the authors let a smart card send the message "Hello World" to another smart card with the help of Amb-BackComs. The experimental results demonstrate that the message can be transmitted with 94% of successful ratio when the two cards are 4 inches apart and communicate at a bit rate of 1 Kbps.

4.6. Logistics

Logistics applications can greatly benefit from ambient backscatter technology as it can help to manage the logistics items. In particular, backscatter tags can be attached to the surface of the goods to assist in reading their information. The information collected by the tag can be reported to the logistics administrator by leveraging nearby ambient signals. In this way, the logistics administrator can manage the items conveniently, thereby improving the management efficiency of logistics items. The work [1] has demonstrated that Amb-BackComs can be adopted to help monitor the items on a shelf.

4.7. Vision applications

Amb-BackComs can also be applied in real-time vision applications such as face detection and recognition. In [83], ambient backscatter technology is implemented to assist in the face detection and recognition process. Specifically, the authors design a novel camera board. The camera board is able to transmit the collected image streaming message to a backscatter receiver (e.g., a laptop) by backscattering Wi-Fi signals. After authenticating the received images, a corresponding action can be performed, such as opening the door for users. The experimental results present a face detection accuracy of 90.1% and a face recognition accuracy of 98.5%.

5. Open issues and challenges

In this section, we discuss some remaining open issues and challenges that are to be addressed to motivate further research efforts in the area of Amb-BackComs.

5.1. Full compatibility with existing infrastructures

Most of the exiting Amb-BackComs systems are not fully compatible with existing infrastructures. Either a specific sequence of excitation signals is required, or deployment of a dedicated receiver is required. The studies in [26,28–31] require single-tone Bluetooth transmissions for backscatter. As a result, excitation signals in these studies cannot be used for productive data communication and the carrier at the same time, thereby greatly reducing spectrum efficiency. The studies in [36,39,60,62,63,79] require modifications to commodity devices or deployment of a dedicated receiver, thereby limiting the commodity device usage and causing more cost. Designing an Amb-BackComs system that is fully compatible with existing infrastructures is challenging since existing devices are specifically designed to receive signals from a specific source. It is unclear how these devices would decode other types of signals from the backscatter tag. The studies in [22,67,68] show full compatibility with existing Wi-Fi infrastructures and the work in [1] show full compatibility with existing TV infrastructures. However, communication performance is limited in terms of throughput and communication range. Thus, further efforts are needed to improve throughput and communication range.

5.2. Interference to existing systems

Ambient backscatter leverages signals transmitted in existing systems. As a result, the reflected signals can cause interference to existing systems. In [1], the authors evaluated the effect of the backscatter on the existing TV receiver and showed that, for the backscatter rates less than 10 Kbps, the backscattering does not lead to a noticeable interference to the TV receiver unless its distance to the tag is less than 7.2 inches. In addition, recent studies in [22,68,69] have demonstrated that backscattering does not cause any noticeable interference to Wi-Fi networks. However, this may not be the case for other systems. Thus, more studies need to be done to evaluate the impact of Amb-BackComs on existing systems.

5.3. Downlink communication

Almost all the ambient backscatter researches focus on uplink transmission (i.e., the transmission from the backscatter tag to the backscatter receiver). There exists a huge gap in the exploration of downlink transmission (i.e., the transmission from ambient sources to backscatter tags). Existing backscatter downlink designs leverage the presence and duration time of the excitation signal to encode information, and use a low-power energy detector to identify the presence of a packet and demodulate data. This inevitably leads to low throughput. In addition, the low-power requirement also restricts the use of powerhungry components on the backscatter tag, thereby further degrading the performance of downlink transmission. Therefore, in-depth research in this direction is necessary.

5.4. Enabling multi-hop backscatter networks

Ambient backscatter research remains in its infancy with primary focus on single-hop design. The robustness and scalability of Amb-BackComs can be severely limited by single-hop design owing to poor channel and its restricted connectivity. Recent work [70] has demonstrated that a multi-hop backscatter can provide path diversity and robustness for tag data transmission. However, its performance is limited to two hops, achieving a throughput of only 200 bps with tag-to-tag distances of 40 cm. There is significant research to be done on multihop design to extend the coverage of Amb-BackComs systems as well as achieve higher data rates and longer communication ranges.

5.5. Medium access control protocol

The practical application of Amb-BackComs involves the communication of numerous tags. These tags are deployed in our surrounding environment to perform a wide range of activities (e.g., gas leakage detection) and could collaborate to send the collected data to the receiver employing a single-hop or multi-hop mechanism. An efficient MAC protocol is needed to coordinate multiple tags or support concurrent transmission. Through efficient MAC schemes, the wireless medium between multiple tags can be efficiently shared. Thus, the interference and collision between multi-tag transmission could be handled and suppressed, thereby achieving significant performance gain. FreeRider [22] achieves an aggregate throughput of 15 Kbps when communicating with 20 tags by implementing a MAC scheme. P^2LoRa [84] achieves the parallel transmission of up to 101 tags with ambient LoRa as the excitation signal. NetScatter [85] provides a throughput of about 250 Kbps when communicating with up to 256 concurrent tags using distributed chirp spread spectrum coding, but only works with LoRa signals. Therefore, further efforts are needed to support large-scale deployment regardless of ambient signals.

5.6. Full-duplex technique

Declaration of competing interest

Computer Networks 216 (2022) 109235

Amb-BackComs can benefit from the full-duplex technique. With the full-duplex operation, the backscatter tag can transmit and receive data at the same time and frequency. Thus, the performance of Amb-BackComs systems can be improved including, but not limited to transmission latency, spectral efficiency, and throughput. Achieving full-duplex communication on the low-power tag is challenging since traditional full-duplex techniques usually require complex hardware design, which is power-intensive. Prior studies [86,87] have demonstrated the full-duplex communication on the low-power tag. However, these methods suffer from interference and spectrum efficiency. Therefore, further researches in this direction are needed to solve these problems.

5.7. Channel capacity and coverage analysis

Channel capacity and coverage are important indicators of system performance. Owing to the low-power nature of the backscatter tag as well as the unpredictability and burstiness of ambient signals, traditional models of coverage and capacity analysis cannot be directly applied into ambient backscatter systems. Recent studies [88,89] have taken a step forward in this paradigm. Zhao et al. [88] investigated the capacity problem of an ambient backscatter system by using the Gaussian channel model. Wei et al. [89] performed the capacity analysis of an 802.11b-based ambient backscatter system. These studies open the door for further efforts in this direction.

6. Conclusion

Ambient backscatter technology is promising to play a vital role in future low-power communications systems, such as IoT. Further efforts in this direction are needed to enable a practical ambient backscatter system that is sufficiently effective for real-world IoT applications. In this paper, we have provided a detailed discussion on different Amb-BackComs systems that have been reported in the literature between 2013 and 2021. We believe that new Amb-BackComs systems with full compatibility with existing infrastructures and higher performance will be an interesting topic in the following years. The evaluation and analysis of the interference to existing systems is also an important research topic. Also, the solutions for enhancing the performance of downlink communication in Amb-BackComs systems will be a hot direction. Another interesting future topic is the design of a practical multi-hop backscatter system. Considering that future IoT applications comprise numerous backscatter tags, the design of an efficient MAC protocol is also a hot research direction. Apart from this, full-duplex communication on the tag is still a challenging topic, because traditional full-duplex communication generally requires power-hungry components, which are not feasible for low-power tags. In addition, coverage and capacity analysis in Amb-BackComs systems will be good directions.

Furthermore, this paper provided a detailed discussion on basic principles and potential applications. We believe that this paper is able to provide readers, interested in Amb-BackComs, with a comprehensive and detailed insight into different aspects of Amb-BackComs, and that this paper is expected to act as a starting point for future research.

CRediT authorship contribution statement

Weiqi Wu: Writing – original draft. Xingfu Wang: Supervision, Writing – review and editing, Resources. Ammar Hawbani: Supervision, Writing – review and editing, Resources. Longzhi Yuan: Resources. Wei Gong: Supervision, Writing – review and editing, Resources. The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

References

- Vincent Liu, Aaron Parks, Vamsi Talla, Shyamnath Gollakota, David Wetherall, Joshua R. Smith, Ambient backscatter: Wireless communication out of thin air, ACM SIGCOMM Comput. Commun. Rev. 43 (4) (2013) 39–50.
- [2] Taekyung Kim, Wonjun Lee, Anyscatter: Eliminating technology dependency in ambient backscatter systems, in: IEEE INFOCOM 2020-IEEE Conference on Computer Communications, IEEE, 2020, pp. 287–296.
- [3] Ander Galisteo, Ambuj Varshney, Domenico Giustiniano, Two to tango: Hybrid light and backscatter networks for next billion devices, in: Proceedings of the 18th International Conference on Mobile Systems, Applications, and Services, 2020, pp. 80–93.
- [4] John Kimionis, Aggelos Bletsas, John N. Sahalos, Increased range bistatic scatter radio, IEEE Trans. Commun. 62 (3) (2014) 1091–1104.
- [5] Xiao Lu, Ping Wang, Dusit Niyato, Dong In Kim, Zhu Han, Wireless charging technologies: Fundamentals, standards, and network applications, IEEE Commun. Surv. Tutor. 18 (2) (2015) 1413–1452.
- [6] Shin Hyuk Choi, Dong In Kim, Backscatter radio communication for wireless powered communication networks, in: 2015 21st Asia-Pacific Conference on Communications (APCC), IEEE, 2015, pp. 370–374.
- [7] Wei Zhang, Yao Qin, Wenjing Zhao, Minzheng Jia, Qiang Liu, Ruisi He, Bo Ai, A green paradigm for Internet of Things: Ambient backscatter communications, China Commun. 16 (7) (2019) 109–119.
- [8] Abhishek Khanna, Sanmeet Kaur, Internet of things (IoT), applications and challenges: A comprehensive review, Wirel. Pers. Commun. 114 (2020) 1687–1762.
- [9] Laphou Lao, Zecheng Li, Songlin Hou, Bin Xiao, Songtao Guo, Yuanyuan Yang, A survey of IoT applications in blockchain systems: Architecture, consensus, and traffic modeling, ACM Comput. Surv. 53 (1) (2020) 1–32.
- [10] Usman Saleh Toro, Kaishun Wu, Victor C.M. Leung, Backscatter wireless communications and sensing in green internet of things, IEEE Trans. Green Commun. Netw. (2021).
- [11] Nguyen Van Huynh, Dinh Thai Hoang, Xiao Lu, Dusit Niyato, Ping Wang, Dong In Kim, Ambient backscatter communications: A contemporary survey, IEEE Commun. Surv. Tutor. 20 (4) (2018) 2889–2922.
- [12] Chenren Xu, Lei Yang, Pengyu Zhang, Practical backscatter communication systems for battery-free Internet of Things: A tutorial and survey of recent research, IEEE Signal Process. Mag. 35 (5) (2018) 16–27.
- [13] Mudasar Latif Memon, Navrati Saxena, Abhishek Roy, Dong Ryeol Shin, Backscatter communications: Inception of the battery-free era—A comprehensive survey, Electronics 8 (2) (2019) 129.
- [14] Ruifeng Duan, Xiyu Wang, Huseyin Yigitler, Muhammad Usman Sheikh, Riku Jantti, Zhu Han, Ambient backscatter communications for future ultra-low-power machine type communications: Challenges, solutions, opportunities, and future research trends, IEEE Commun. Mag. 58 (2) (2020) 42–47.
- [15] Tushar S. Muratkar, Ankit Bhurane, Ashwin Kothari, Battery-less internet of things-a survey, Comput. Netw. 180 (2020) 107385.
- [16] Fatemeh Rezaei, Chintha Tellambura, Sanjeewa Herath, Large-scale wirelesspowered networks with backscatter communications—A comprehensive survey, IEEE Open J. Commun. Soc. 1 (2020) 1100–1130.
- [17] Wanchun Liu, Kaibin Huang, Xiangyun Zhou, Salman Durrani, Next generation backscatter communication: systems, techniques, and applications, EURASIP J. Wireless Commun. Networking 2019 (1) (2019) 1–11.
- [18] Xiao Lu, Dusit Niyato, Hai Jiang, Dong In Kim, Yong Xiao, Zhu Han, Ambient backscatter assisted wireless powered communications, IEEE Wirel. Commun. 25 (2) (2018) 170–177.
- [19] Dong Ma, Guohao Lan, Mahbub Hassan, Wen Hu, Sajal K. Das, Sensing, computing, and communications for energy harvesting IoTs: A survey, IEEE Commun. Surv. Tutor. 22 (2) (2019) 1222–1250.
- [20] Jia Zhao, Wei Gong, Jiangchuan Liu, Spatial stream backscatter using commodity wifi, in: Proceedings of the 16th Annual International Conference on Mobile Systems, Applications, and Services, 2018, pp. 191–203.
- [21] Yao Peng, Longfei Shangguan, Yue Hu, Yujie Qian, Xianshang Lin, Xiaojiang Chen, Dingyi Fang, Kyle Jamieson, PLoRa: A passive long-range data network from ambient LoRa transmissions, in: Proceedings of the 2018 Conference of the ACM Special Interest Group on Data Communication, 2018, pp. 147–160.
- [22] Pengyu Zhang, Colleen Josephson, Dinesh Bharadia, Sachin Katti, Freerider: Backscatter communication using commodity radios, in: Proceedings of the 13th International Conference on Emerging Networking EXperiments and Technologies, 2017, pp. 389–401.
- [23] Vamsi Talla, Joshua R. Smith, Hybrid analog-digital backscatter: A new approach for battery-free sensing, in: 2013 IEEE International Conference on RFID (RFID), IEEE, 2013, pp. 74–81.

- [24] M. Katanbaf, V. Jain, Joshua R. Smith, Relacks: Reliable backscatter communication in indoor environments, Proc. ACM Interact. Mob. Wearable Ubiquitous Technol. 4 (2020) 48:1–48:24.
- [25] [n. d.]. Bluetooth Core Specification, 2019, https://www.bluetooth.com/ specifications/bluetooth-core-specification/.
- [26] Maolin Zhang, Jia Zhao, Si Chen, Wei Gong, Reliable backscatter with commodity BLE, in: IEEE INFOCOM 2020-IEEE Conference on Computer Communications, IEEE, 2020, pp. 1291–1299.
- [27] Joshua F. Ensworth, Matthew S. Reynolds, Ble-backscatter: Ultralow-power IoT nodes compatible with Bluetooth 4.0 low energy (BLE) smartphones and tablets, IEEE Trans. Microw. Theory Tech. 65 (9) (2017) 3360–3368.
- [28] Vikram Iyer, Vamsi Talla, Bryce Kellogg, Shyamnath Gollakota, Joshua Smith, Inter-technology backscatter: Towards internet connectivity for implanted devices, in: Proceedings of the 2016 ACM SIGCOMM Conference, 2016, pp. 356–369.
- [29] Maolin Zhang, Si Chen, Amiya Nayak, Wei Gong, Enabling multi-channel backscatter communication for bluetooth low energy, in: ICC 2020-2020 IEEE International Conference on Communications (ICC), IEEE, 2020, pp. 1–6.
- [30] Si Chen, Maolin Zhang, Jia Zhao, Wei Gong, Jiangchuan Liu, Reliable and practical bluetooth backscatter with commodity devices, IEEE/ACM Trans. Netw. (2021).
- [31] Maolin Zhang, Si Chen, Jia Zhao, Wei Gong, Commodity-level BLE backscatter, in: Proceedings of the 19th Annual International Conference on Mobile Systems, Applications, and Services, 2021, pp. 402–414.
- [32] ZigBee Alliance, ZigBee Document 05-3474-21, https://zigbeealliance.org/wpcontent/uploads/2019/11/docs-05-3474-21-0csg-zigbee-specification.pdf.
- [33] IEEE Standards Association, et al., IEEE std 802.15. 4-2011, IEEE standard for local and metropolitan area networks—part 15.4: Low-rate wireless personal area networks (LR-WPANs), 2011.
- [34] Wei Gong, Longzhi Yuan, Qiwei Wang, Jia Zhao, Multiprotocol backscatter for personal IoT sensors, in: Proceedings of the 16th International Conference on Emerging Networking Experiments and Technologies, 2020.
- [35] Advanced Television Systems Committee, et al., A/53: ATSC digital television standard, parts 1-6, XP-008115585 (2007) 1–136.
- [36] Aaron N. Parks, Angli Liu, Shyamnath Gollakota, Joshua R. Smith, Turbocharging ambient backscatter communication, ACM SIGCOMM Comput. Commun. Rev. 44 (4) (2014) 619–630.
- [37] ATS Committee, et al., Atsc recommended practice: Receiver performance guidelines, Tech. Rep. Document A/74: 2010, 2010.
- [38] Anran Wang, Vikram Iyer, Vamsi Talla, Joshua R. Smith, Shyamnath Gollakota, {FM} backscatter: Enabling connected cities and smart fabrics, in: 14th {USENIX} Symposium on Networked Systems Design and Implementation ({NSDI} 17), 2017, pp. 243–258.
- [39] Spyridon Nektarios Daskalakis, John Kimionis, Ana Collado, George Goussetis, Manos M. Tentzeris, Apostolos Georgiadis, Ambient backscatterers using FM broadcasting for low cost and low power wireless applications, IEEE Trans. Microw. Theory Tech. 65 (12) (2017) 5251–5262.
- [40] [n. d.]. LoRa Alliance, https://lora-alliance.org/.
- [41] Fusang Zhang, Zhaoxin Chang, Kai Niu, Jie Xiong, Beihong Jin, Qin Lv, Daqing Zhang, Exploring lora for long-range through-wall sensing, Proc. ACM Interact., Mob., Wearable Ubiquitous Technol. 4 (2) (2020) 1–27.
- [42] Aloÿs Augustin, Jiazi Yi, Thomas Clausen, William Mark Townsley, A study of LoRa: Long range & low power networks for the internet of things, Sensors 16 (9) (2016) 1466.
- [43] Vamsi Talla, Mehrdad Hessar, Bryce Kellogg, Ali Najafi, Joshua R. Smith, Shyamnath Gollakota, Lora backscatter: Enabling the vision of ubiquitous connectivity, Proc. ACM Interact., Mob., Wearable Ubiquitous Technol. 1 (3) (2017) 1–24.
- [44] Albert Berni, W.O. Gregg, On the utility of chirp modulation for digital signaling, IEEE Trans. Commun. 21 (6) (1973) 748–751.
- [45] Xiuzhen Guo, Longfei Shangguan, Yuan He, Jia Zhang, Haotian Jiang, Awais Ahmad Siddiqi, Yunhao Liu, Aloba: Rethinking ON-OFF keying modulation for ambient lora backscatter, in: Proceedings of the 18th Conference on Embedded Networked Sensor Systems, 2020, pp. 192–204.
- [46] IEEE standard for information technology-telecommunications and information exchange between systems - local and metropolitan area networks-specific requirements - part 11: Wireless LAN medium access control (MAC) and physical layer (PHY) specifications - redline, IEEE Std 802.11-2020 (Revision of IEEE Std 802.11-2016) - Redline, 2021, pp. 1–7524.
- [47] Hassan Aboubakr Omar, Khadige Abboud, Nan Cheng, Kamal Rahimi Malekshan, Amila Tharaperiya Gamage, Weihua Zhuang, A survey on high efficiency wireless local area networks: Next generation WiFi, IEEE Commun. Surv. Tutor. 18 (4) (2016) 2315–2344.
- [48] Xiping Wu, Mohammad Dehghani Soltani, Lai Zhou, Majid Safari, Harald Haas, Hybrid LiFi and WiFi networks: A survey, IEEE Commun. Surv. Tutor. 23 (2) (2021) 1398–1420.
- [49] Faheem Zafari, Athanasios Gkelias, Kin K. Leung, A survey of indoor localization systems and technologies, IEEE Commun. Surv. Tutor. 21 (3) (2019) 2568–2599.
- [50] IEEE standard for telecommunications and information exchange between systems - LAN/MAN specific requirements - part 11: Wireless medium access control (MAC) and physical layer (PHY) specifications: High speed physical layer in the 5 GHz band, IEEE Std 802.11a-1999, 1999, pp. 1–102.

- [51] IEEE standard for information technology telecommunications and information exchange between systems - local and metropolitan networks - specific requirements - part 11: Wireless LAN medium access control (MAC) and physical layer (PHY) specifications: Higher speed physical layer (PHY) extension in the 2.4 GHz band, IEEE Std 802.11b-1999, 2000, pp. 1–96.
- [52] IEEE standard for information technology- local and metropolitan area networks-specific requirements- part 11: Wireless LAN medium access control (MAC) and physical layer (PHY) specifications: Further higher data rate extension in the 2.4 GHz band, IEEE Std 802.11g-2003 (Amendment To IEEE Std 802.11, 1999 Edn. (Reaff 2003) As Amended By IEEE Stds 802.11a-1999, 802.11b-1999/Cor 1-2001, and 802.11d-2001), 2003, pp. 1–104.
- [53] IEEE standard for information technology– local and metropolitan area networks-specific requirements– part 11: Wireless LAN medium access control (MAC)and physical layer (PHY) specifications amendment 5: Enhancements for higher throughput, IEEE Std 802.11n-2009 (Amendment To IEEE Std 802.11-2007 As Amended By IEEE Std 802.11k-2008, IEEE Std 802.11r-2008, IEEE Std 802.11y-2008, and IEEE Std 802.11w-2009), 2009, pp. 1–565.
- [54] Lu Lu, Geoffrey Ye Li, A. Lee Swindlehurst, Alexei Ashikhmin, Rui Zhang, An overview of massive MIMO: Benefits and challenges, IEEE J. Sel. Top. Sign. Proces. 8 (5) (2014) 742–758.
- [55] Xin Liu, Zicheng Chi, Wei Wang, Yao Yao, Ting Zhu, Vmscatter: A versatile {MIMO} backscatter, in: 17th {USENIX} Symposium on Networked Systems Design and Implementation ({NSDI} 20), 2020, pp. 895–909.
- [56] [n. d.]. 3GPP Release 8, 2008, https://www.3gpp.org/specifications/releases/72release-8.
- [57] H. Mousavi, Iraj S. Amiri, M.A. Mostafavi, C.Y. Choon, Lte physical layer: Performance analysis and evaluation, Appl. Comput. Inform. 15 (1) (2019) 34–44.
- [58] Zicheng Chi, Xin Liu, Wei Wang, Yao Yao, Ting Zhu, Leveraging ambient lte traffic for ubiquitous passive communication, in: Proceedings of the Annual Conference of the ACM Special Interest Group on Data Communication on the Applications, Technologies, Architectures, and Protocols for Computer Communication, 2020, pp. 172–185.
- [59] [n. d.]. Bluetooth Low energy over-the-air advertiser testing, 2020, https://www.rohde-schwarz.com/hk/applications/bluetooth-low-energy-overthe-air-advertiser-testing-application-note_56280-1001743.html.
- [60] Spyridon-Nektarios Daskalakis, John Kimionis, Ana Collado, Manos M. Tentzeris, Apostolos Georgiadis, Ambient FM backscattering for smart agricultural monitoring, in: 2017 IEEE MTT-S International Microwave Symposium (IMS), IEEE, 2017, pp. 1339–1341.
- [61] Valeria Palazzari, Paolo Mezzanotte, Federico Alimenti, Francesco Fratini, Giulia Orecchini, Luca Roselli, Leaf compatible "eco-friendly" temperature sensor clip for high density monitoring wireless networks, Wirel. Power Transf. 4 (1) (2017) 55–60.
- [62] Georgios Vougioukas, Aggelos Bletsas, 24µ Watt 26 m range batteryless backscatter sensors with FM remodulation and selection diversity, in: 2017 IEEE International Conference on RFID Technology & Application (RFID-TA), IEEE, 2017, pp. 237–242.
- [63] Georgios Vougioukas, Aggelos Bletsas, Switching frequency techniques for universal ambient backscatter networking, IEEE J. Sel. Areas Commun. 37 (2) (2018) 464–477.
- [64] Georgios Vougioukas, Panos N. Alevizos, Aggelos Bletsas, Coherent detector for pseudo-FSK backscatter under ambient constant envelope illumination, in: 2018 IEEE 19th International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), IEEE, 2018, pp. 1–5.
- [65] Xinyu Zhang, Kang G. Shin, E-mili: Energy-minimizing idle listening in wireless networks, IEEE Trans. Mob. Comput. 11 (9) (2012) 1441–1454.
- [66] Xiuzhen Guo, Longfei Shangguan, Yuan He, Jia Zhang, Haotian Jiang, Awais Ahmad Siddiqi, Yunhao Liu, Efficient ambient LoRa backscatter with on-off keying modulation, IEEE/ACM Trans. Netw. (2021).
- [67] Bryce Kellogg, Aaron Parks, Shyamnath Gollakota, Joshua R. Smith, David Wetherall, Wi-Fi backscatter: Internet connectivity for RF-powered devices, in: Proceedings of the 2014 ACM Conference on SIGCOMM, 2014, pp. 607–618.
- [68] Pengyu Zhang, Dinesh Bharadia, Kiran Joshi, Sachin Katti, Hitchhike: Practical backscatter using commodity wifi, in: Proceedings of the 14th ACM Conference on Embedded Network Sensor Systems CD-ROM, 2016, pp. 259–271.
- [69] Qiwei Wang, Si Chen, Jia Zhao, Wei Gong, Rapidrider: Efficient WiFi backscatter with uncontrolled ambient signals, in: IEEE INFOCOM 2021-IEEE Conference on Computer Communications, IEEE, 2021, pp. 1–10.
- [70] Jia Zhao, Wei Gong, Jiangchuan Liu, X-tandem: Towards multi-hop backscatter communication with commodity wifi, in: Proceedings of the 24th Annual International Conference on Mobile Computing and Networking, 2018, pp. 497–511.
- [71] Manideep Dunna, Miao Meng, Po-Han Wang, Chi Zhang, Patrick P. Mercier, Dinesh Bharadia, Syncscatter: Enabling WiFi like synchronization and range for WiFi backscatter communication, in: NSDI, 2021, pp. 923–937.
- [72] Xin Liu, Zicheng Chi, Wei Wang, Yao Yao, Pei Hao, Ting Zhu, Verification and redesign of {OFDM} backscatter, in: 18th {USENIX} Symposium on Networked Systems Design and Implementation ({NSDI} 21), 2021, pp. 939–953.

- [73] Ali Abedi, Mohammad Hossein Mazaheri, Omid Abari, Tim Brecht, Witag: Rethinking backscatter communication for wifi networks, in: Proceedings of the 17th ACM Workshop on Hot Topics in Networks, 2018, pp. 148–154.
- [74] Ali Abedi, Farzan Dehbashi, Mohammad Hossein Mazaheri, Omid Abari, Tim Brecht, WiTAG: Seamless WiFi backscatter communication, in: Proceedings of the Annual Conference of the ACM Special Interest Group on Data Communication on the Applications, Technologies, Architectures, and Protocols for Computer Communication, 2020, pp. 240–252.
- [75] Xin He, Weiwei Jiang, Meng Cheng, Xiaobo Zhou, Panlong Yang, Brian Kurkoski, Guardrider: Reliable WiFi backscatter using reed-Solomon codes with QoS guarantee, in: 2020 IEEE/ACM 28th International Symposium on Quality of Service (IWQoS), IEEE, 2020, pp. 1–10.
- [76] Pengyu Zhang, Mohammad Rostami, Pan Hu, Deepak Ganesan, Enabling practical backscatter communication for on-body sensors, in: Proceedings of the 2016 ACM SIGCOMM Conference, 2016, pp. 370–383.
- [77] Jinhwan Jung, Jihoon Ryoo, Yung Yi, Song Min Kim, Gateway over the air: towards pervasive internet connectivity for commodity IoT, in: Proceedings of the 18th International Conference on Mobile Systems, Applications, and Services, 2020, pp. 54–66.
- [78] Zerina Kapetanovic, Ali Saffari, Ranveer Chandra, Joshua R. Smith, Glaze: Overlaying occupied spectrum with downlink iot transmissions, Proc. ACM Interact., Mob., Wearable Ubiquitous Technol. 3 (4) (2019) 1–21.
- [79] Chouchang Yang, Jeremy Gummeson, Alanson Sample, Riding the airways: Ultra-wideband ambient backscatter via commercial broadcast systems, in: IEEE INFOCOM 2017-IEEE Conference on Computer Communications, IEEE, 2017, pp. 1–9.
- [80] Anandghan Waghmare, Qiuyue Xue, Dingtian Zhang, Yuhui Zhao, Shivan Mittal, Nivedita Arora, Ceara Byrne, Thad Starner, Gregory D. Abowd, Ubiquitouch: Self sustaining ubiquitous touch interfaces, Proc. ACM Interact., Mob., Wearable Ubiquitous Technol. 4 (1) (2020) 1–22.
- [81] Manikanta Kotaru, Pengyu Zhang, Sachin Katti, Localizing low-power backscatter tags using commodity wifi, in: Proceedings of the 13th International Conference on Emerging Networking EXperiments and Technologies, 2017, pp. 251–262.
- [82] Elahe Soltanaghaei, Adwait Dongare, Akarsh Prabhakara, Swarun Kumar, Anthony Rowe, Kamin Whitehouse, Tagfi: Locating ultra-low power WiFi tags using unmodified WiFi infrastructure, Proc. ACM Interact., Mob., Wearable Ubiquitous Technol. 5 (1) (2021) 1–29.
- [83] Colleen Josephson, Lei Yang, Pengyu Zhang, Sachin Katti, Wireless computer vision using commodity radios, in: 2019 18th ACM/IEEE International Conference on Information Processing in Sensor Networks (IPSN), IEEE, 2019, pp. 229–240.
- [84] Jinyan Jiang, Zhenqiang Xu, Fan Dang, Jiliang Wang, Long-range ambient LoRa backscatter with parallel decoding, in: Proceedings of the 27th Annual International Conference on Mobile Computing and Networking, 2021, pp. 684–696.
- [85] Mehrdad Hessar, Ali Najafi, Shyamnath Gollakota, Netscatter: Enabling largescale backscatter networks, in: 16th {USENIX} Symposium on Networked Systems Design and Implementation ({NSDI} 19), 2019, pp. 271–284.
- [86] Wanchun Liu, Kaibin Huang, Xiangyun Zhou, Salman Durrani, Full-duplex backscatter interference networks based on time-hopping spread spectrum, IEEE Trans. Wireless Commun. 16 (7) (2017) 4361–4377.
- [87] Vincent Liu, Vamsi Talla, Shyamnath Gollakota, Enabling instantaneous feedback with full-duplex backscatter, in: Proceedings of the 20th Annual International Conference on Mobile Computing and Networking, 2014, pp. 67–78.
- [88] Wenjing Zhao, Gongpu Wang, Feifei Gao, Yulong Zou, Saman Atapattu, Channel capacity and lower bound for ambient backscatter communication systems, in: 2017 9th International Conference on Wireless Communications and Signal Processing (WCSP), IEEE, 2017, pp. 1–6.
- [89] Jiamei Wei, Xiaohua Tian, Capacity analysis of frequency shift based backscatter communication system, in: 2018 3rd International Conference on Computer and Communication Systems (ICCCS), IEEE, 2018, pp. 353–357.



Weiqi Wu received the B.S. degree from the Department of Computer Science and Technology, Anhui University of Science and Technology (AUST). He is currently pursuing the Ph.D. degree with the Department of Computer Science and Technology, University of Science and Technology of China. His research interests include ambient backscatter communication and wireless sensor networks.



Xingfu Wang (Member IEEE) received the B.S. degree in electronic and information engineering from Beijing Normal University of China in 1988, and the M.S. degree in computer science from the University of Science and Technology of China in 1997. He is an associate professor in the School of Compute Science and Technology, University of Science and Technology of China. His current research interests include Information Security, Data Management and WSN.



Ammar Hawbani received the B.S., M.S. and Ph.D. degrees in Computer Software and Theory from the University of Science and Technology of China (USTC), Hefei, China, in 2009, 2012 and 2016, respectively. Currently, he is a postdoctoral researcher in the School of Computer Science and Technology at USTC. His research interests include IoT, WSNs, WBANs, WMNs, VANETs and SDN.



Longzhi Yuan (Student Member, IEEE) received the B.S. degree from the School of Information Science and Technology, University of Science and Technology of China, Hefei, China, in 2017. He is currently pursuing the Ph.D. degree with the School of Data Science, University of Science and Technology of China. His research interests include wireless networks and the IoT applications.



Wei Gong (Member IEEE) received the B.S. degree from the Department of Computer Science and Technology, Huazhong University of Science and Technology, and the M.S. and Ph.D. degrees from the School of Software and Department of Computer Science and Technology, Tsinghua University. He is currently a Professor with the School of Computer Science and Technology, University of Science and Technology of China. His research interests include backscatter networks, edge systems, and the IoT applications.